Contents lists available at ScienceDirect





journal homepage: www.elsevier.com/locate/jempfin



Factor momentum in the Chinese stock market *

Tian Ma^{a,b,#}, Cunfei Liao^{c,#}, Fuwei Jiang^{d,*}

^a School of Economics, Minzu University of China, Beijing 100081, China

^b China Institute for Vitalizing Border Areas and Enriching the People, Beijing, China

^c School of Economics and Management, Nanjing University of Science and Technology, Nanjing 210094, China

^d School of Finance, Central University of Finance and Economics, Beijing 100081, China

ARTICLE INFO

JEL: G12 G14 Keywords: Factor momentum Chinese stock market Mispricing Factor premium Factor timing

ABSTRACT

Based on 10 commonly used factors, we construct a novel factor momentum strategy in the Chinese stock market, which earns an annualized return of 9.91 %, with a Sharpe ratio of 1.15. Factor momentum subsumes stock momentum, high-priced momentum, and industry momentum, digests its component factors and a variety of anomalies, and represents the momentum anomaly in China. Furthermore, mispricing correction helps explain factor momentum, which produces stronger returns during higher aggregate idiosyncratic volatility periods as well as among stocks with higher information asymmetry and short-sale constraints. Exposure to factor premiums and the manifestation of predictability determine factor momentum in China.

1. Introduction

Momentum is widely observed in financial markets such as stock (Jegadeesh and Titman, 1993, 2001), currency (Burnside et al., 2011; Menkhoff et al., 2012), option (Heston et al., 2021), and bond market (Sihvonen, 2021). Recently, factor momentum, where past winners earn higher returns than losers in long-short anomaly portfolio returns, has received much scholarly attention (Arnott et al., 2021; Avramov et al., 2017; Ehsani and Linnainmaa, 2022; Leippold and Yang, 2021; Zaremba and Shemer, 2018). However, no studies have thus far investigated factor momentum in China. Does factor momentum exist in China, where merely generates stock momentum? Do the unique features of the Chinese stock market affect its performance? Our paper expands factor momentum in China's stock market, investigates its explanatory power, and shows its economic sources.

We investigate factor momentum in China for three important reasons. First, China's stock market is the second-largest in the world, and its performance has a considerable influence on global capital markets (Allen et al., 2020). Second, the Chinese stock

* Corresponding author.

E-mail address: jfuwei@gmail.com (F. Jiang).

https://doi.org/10.1016/j.jempfin.2023.101458

Received 25 October 2022; Received in revised form 27 July 2023; Accepted 5 December 2023 Available online 7 December 2023 0927-5398/© 2023 Elsevier B.V. All rights reserved.

^{*} We are grateful to Kewei Hou, anonymous reviewer, Haiqi Li, Yan Liu, Guohao Tang, Yudong Wang, Siyuan Yang (discussant), Xueyong Zhang, seminar participants at Hunan University, and conference participants at the 2022 China International Risk Forum and 2022 SoFiE Financial Econometrics Summer School. Jiang acknowledges the financial support from the National Social Science Fund of China (22&ZD063), National Natural Science Foundation of China (No. 72072193, 71872195) and the Programe for Innovation Research in CUFE. Ma acknowledges the financial support from the National Natural Science Foundation of China (72303271) and the Fundamental Research Funds for the Central Universities (2023QNTS20). Liao acknowledges the financial support from the National Natural Science Foundation of China (30923010409), and the General Project of Philosophy and Social Science Research in Colleges and Universities in Jiangsu Province (2023SJYB0024).

[#] These authors contributed equally to this work and should be considered co-first authors.

market lacks stock-level momentum (e.g., Pan et al., 2013; Gao et al., 2021). Chui et al. (2010) find that momentum is intriguingly absent in the Chinese market, in contrast to its' significantly present in the U.S. market. This difference between the world's two top equity markets attracts much attention from global investors. If China's stock market does indeed deliver significant factor momentum, it would supplement unique evidence on the momentum anomaly. Third, in contrast to other global markets, the Chinese stock market possesses unique features. As one of the important explanations for stock momentum, irrational behavior biases cause stock prices to deviate from their fundamental value (e.g., Daniel et al., 1998; Chui et al., 2010; Grinblatt and Han, 2005; George and Hwang, 2004). The Chinese market, with its high participation of retail investors and market frictions; could provide insights into the economic channels of momentum.

We first examine whether individual factors have time-series momentum (Moskowitz et al., 2012) in China. We use ten mainstream non-momentum factors from the asset pricing literature following Ehsani and Linnainmaa (2022): size (Banz, 1981), value (Rosenberg et al., 1985), profitability (Novy-Marx, 2013), investment (Titman et al., 2004), illiquidity (Amihud, 2002), the earnings-to-price ratio (Basu, 1983), accruals (Sloan, 1996), the cash flow-to-price ratio (Rosenberg et al., 1985), turnover (Datar et al., 1998), and betting against beta (Frazzini and Pedersen, 2014). The findings show that the individual time-series factor momentum strategy, which buys the factor with an above-zero return and sells it with negative performance over the prior year, earns significantly positive returns for most of the factors. Certain factor models, such as Fama and French's (2015) five-factor model (FF5) and Liu et al. (2019) Chinese three-factor model (CH3), cannot explain individual factor momentum.

Furthermore, we construct two aggregately factor momentum strategies. The first one is the time-series factor momentum approach (TSFM; Gupta and Kelly, 2019; Ehsani and Linnainmaa, 2022), which longs factors with positive prior one-year returns and shorts those with negative returns, generating an annualized average return of 9.91 % (t = 6.07) and a Sharpe ratio of 1.15. The second strategy is the cross-sectional factor momentum strategy (CSFM; Avramov et al., 2017; Arnott et al., 2021), which buys factors with above-median returns relative to other factors over the prior year and shorts those with poor performance. The CSFM approach earns a 7.02 % (t = 3.44) average return per annum and a Sharpe ratio of 0.80. Both long-short strategies produce significant abnormal FF5 and CH3 alphas. Notably, we construct a conditional CH3 factor model that imposes Baker and Wurgler's (2006) investor sentiment into the model's coefficients to capture the time-varying components. The conditional CH3 model also generates significantly positive alphas of the factor momentum portfolios. Moreover, the long-leg portfolios generate significantly positive returns, while the short-legs produce insignificant returns. In addition, the TSFM strategy outperforms the CSFM because it purely focuses on the factor auto-correlations without being influenced by cross-serial covariances. Our findings suggest that factors persist and generate momentum in China.

Moreover, we investigate factor momentum's ability to digest stock anomalies, as it is aggregated from individual anomalies. The findings show that the TSFM portfolio has the additional explanatory ability for its component factors and, more generally, 50 fundamental-related anomalies in China than CH3, including value versus growth, profitability, investment, trading friction, and intangibles-related characteristics (e.g., Green et al., 2017; Gu et al., 2020; Hou et al., 2020). Over 80 % of the factors and anomalies produce lower abnormal alphas when adding TSFM as a control variable. The average absolute value of the TSFM augmented with the CH3 alpha is insignificant at 4.48 (t = 1.36), the slope of the TSFM is positively significant at a value of 0.39 (t = 2.94), and the regression R^2 is 46.34 %. Our results highlight the potential application value of factor momentum.

Due to the uniqueness of the insignificant stock momentum in the Chinese market, we investigate the relation between stock and factor momentum. The findings show that factor momentum subsumes the standard stock-level momentum, high-priced stock momentum (Du et al., 2022), and industry momentum and remains positively significant when adjusting for the momentum or reversal strategies. Factor momentum, which is the sum of the autocorrelations in the factors, represents the momentum anomaly in the Chinese stock market. Besides, we investigate whether factor momentum crashes with market states. We observe that factor momentum returns remain robust overall and have asymmetry following market declines. Our findings alleviate concerns that factor momentum strategies experience strings of negative returns, as momentum portfolios often do (Daniel and Moskowitz, 2016).

We proceed to analyze the economic mechanisms underlying factor momentum. First, we find that factor momentum arises from mispricing correction due to slow-moving arbitrage capital and limits to arbitrage. Factor returns form because of mispricing. Arbitragers who want to generate persistent factor returns trade gradually to correct this mispricing instead of aggressively eliminating all mispricing. According to Abreu and Brunnermeier (2002), there is delayed arbitrage because of the synchronization risk, the uncertainty about when peers of arbitragers take advantage of a common arbitrage opportunity. Then, during the high limits-to-arbitrage periods, arbitrageurs are more likely to take participation in trading due to market frictions (Li et al., 2022), resulting in higher returns of factor momentum. Our results demonstrate that factor momentum strengthens when there is high aggregate idiosyncratic volatility, low investor sentiment, high information asymmetry, and short-sale constraints, supporting the hypothesis that when limits to arbitrage exist, increased arbitrageur participation is associated with stronger factor momentum. Second, we decompose the factor momentum strategy into the factor timing and buy-and-hold portfolios (Leippold and Yang, 2021) and find that factor momentum is a manifestation of factor timing and exposure to factor premiums in China. Timing is important in the Chinese market, due to the high standard deviations of the factor returns, the large proportions of retail investors, and the variability of market states, compared to the findings of Leippold and Yang (2021) in the U.S. Third, our paper documents the impact of culture on factor momentum. Collectivistic culture markets' investors rely more on collective intelligence, leading to herding and intensification of the co-movement and autocorrelations between factors. Consequently, a collectivistic market like China lacks stock-level momentum (Chui et al., 2010) but generates significant factor momentum.

In the robustness checks, we first investigate how factor momentum performs after changing the formation and holding periods and find that it persists. Second, we construct factor momentum using an expanded 60-anomaly set and find that it is insensitive to the choice of anomaly set. Third, we examine factor momentum performance during two stock market crash periods in China and find that

T. Ma et al.

the results are similar to those of the full sample. Fourth, we find that factor momentum remains robust despite considering the price limit in china.¹

Our paper contributes to studies of factor momentum in two main ways. First, we find that factor momentum is economically large and statistically significant in the Chinese stock market, despite the lack of stock-level momentum. Our robust results supplement the investigation that factor momentum is distinct from individual momentum and concur with previous findings on factor momentum and factor timing (Ehsani and Linnainmaa, 2022; Haddad et al., 2020; Leippold and Yang, 2021). Second, we provide a more comprehensive understanding of the linkages between time-series and cross-sectional studies than previous research has offered (Dong et al., 2022; Engelberg et al., 2021). We demonstrate that factor momentum can explain stock momentum, its component factors, and a large number of anomalies. Notably, we identify the potential channels through which factor momentum operates, such as the correction of mispricing. Zhang (2022) argues that limits-to-arbitrage help explain the carry factor momentum, as the currencies of the developed markets, countries with low capital account openness, and those with high idiosyncratic risk and currency volatility, experience higher factor momentum. Li et al. (2022) suggest that the level of arbitrageur participation and limits to arbitrage have an impact on the risk-based return momentum. Our paper is related to Zhang (2022) and Li et al. (2022) but extends the mispricing correction within the field of factor momentum. In addition, For the latter, our approach is the first to illustrate the cultural impact on the factor momentum, in the spirit of Chui et al. (2010).

Our study also expands the body of knowledge by providing novel evidence for China. First, we find that the highest returngenerating strategy is related to one-year formation periods in China, whereas shorter formation periods perform better in most other global markets (Gupta and Kelly, 2019). This difference may stem from the high participation of short-horizon retail investors in China (Jones et al., 2020), which causes significant short-term reversals (Du et al., 2022) and attenuates factor momentum. Second, our results underscore the significant contribution of timing in China's stock market (Leippold et al., 2021; Tang et al., 2021), which differs from Leippold and Yang's (2021) findings in the U.S. The reason for this difference may be due to the larger variances of factor returns, the higher ratio of retail investors in the Chinese market who have more severe behavioral biases than institutional investors, and the collectivistic culture of the market that leads to herding. Third, the market structure of China, such as short-sale constraints, is essential in decomposing the returns of factor momentum.

The remainder of this paper is organized as follows. Section II describes the data. Section III presents the empirical results of factor momentum in the Chinese stock market, including its power in explaining anomalies. Section IV investigates the stock momentum and factor momentum. Section V describes the economic channels of factor momentum. Section VI presents the robustness tests. Section VII concludes.

2. Data

We obtain data from the China Stock Market & Accounting Research database from January 2001 to December 2019, including financial statement data and monthly and daily stock returns for all A-share stocks listed on the Shanghai and Shenzhen Stock Exchanges. To ensure the quality of the data, we exclude stocks with special and/or particular transfer status, as these tend to be under financial distress (Carpenter et al., 2021)).

We construct 10 characteristic-based non-momentum factor portfolios following Ehsani and Linnainmaa (2022) and Liu et al. (2019): size (Banz, 1981), value (Rosenberg et al., 1985), profitability (Novy-Marx, 2013), investment (Titman et al., 2004), illiquidity (Amihud, 2002), the earnings-to-price ratio (Basu, 1983), accruals (Sloan, 1996), the cash flow-to-price ratio (Rosenberg et al., 1985), turnover (Datar et al., 1998), and betting against beta (Frazzini and Pedersen, 2014).² These 10 non-momentum factors are the most commonly used in the global market. We calculate these factors following the original studies, with the annualized factor return defined as the difference between the average return in the top quintile and that in the bottom quintile.³

Panel A of Table 1 shows the significant variation in average annualized returns. For example, the earnings-to-price ratio is 14.33 % with a *t*-statistic of 4.78 in China. The standard deviations of the factors also vary significantly, ranging from 10.48 for accruals to 31.06 for betting against beta, indicating their high volatility. The average returns of the factors in China are higher than those in the United States (Liu et al., 2019), and the standard deviations.

Panel B of Table 1 reports the Pearson correlations of the 10 non-momentum factors. Most factor pairs are correlated and below 0.50 in absolute value. These 10 factors thus contain distinct return behavior and represent large and well-diversified portfolios.

3. Factor momentum

3.1. Individual factor momentum

We first investigate whether each factor individually generates factor momentum. The time series momentum methodology is based on the idea of "portfolio timing" (Moskowitz et al., 2012). For each month, we long the factor with positive returns over the prior

¹ China's equity market imposes daily price limits of 10% on regular stocks.

 $^{^2}$ We do not include momentum-related factors because these are considered insignificant in the Chinese stock market.

³ Online Appendix B provides the detailed construction of the ten factors.

Descriptive Statistics.

Panel A reports the average annualized returns (in percentages), *t*-statistics (in square brackets), and standard deviations for the 10 non-momentum factors in China based on the literature. The factor return is defined as the high-quintile return minus the low-quintile return. Panel B reports the correlation matrix of the 10 non-momentum factors. The sample period runs from January 2002 to December 2019.

Panel A: Des	criptive Statist	ics								
Factor			Literature	:			M	ean	STD	
Size (SIZE)			Banz (198	Banz (1981)				0.02 [1.76]	24.86	
Value (BM)			Rosenberg	g et al. (1985)			17	.84 [4.27]	18.21	
Profitability	(GP)		Novy-Mar	x (2013)			12	.48 [2.73]	19.95	
Investment (CINVEST)		Titman et	al. (2004)			8.	91 [2.66]	14.61	
Liquidity (ILl	L)		Amihud (2002)			9.	70 [2.07]	20.44	
Earnings to p	rice ratio (EP))	Basu (198	33)			14	.33 [4.78]	13.05	
Accruals (AC	C)		Sloan (19	96)			11	.17 [4.65]	10.48	
Cash-flow to	price (CFP)		Rosenberg	g et al. (1985)			1.	69 [0.46]	15.82	
Turnover (Tl	JRN)		Datar et a	l. (1998)			12	2.38 [3.59]	15.04	
Betting-agair	st-beta (BAB)		Frazzini a	nd Pedersen (201	4)		11	.43 [1.60]	31.06	
Panel B: Corr	elation Matrix	C								
	SIZE	BM	GP	CINVEST	ILL	EP	ACC	CFP	TURN	BAB
SIZE	1.00	0.43	-0.47	0.04	0.94	-0.67	-0.03	0.69	0.50	-0.08
BM		1.00	0.27	0.26	0.35	-0.31	0.50	0.57	0.36	0.13
GP			1.00	0.47	-0.54	0.63	0.43	-0.45	-0.43	0.40
CINVEST				1.00	0.01	0.04	-0.08	-0.25	-0.08	0.27
ILL					1.00	-0.64	-0.07	0.65	0.46	-0.09
EP						1.00	0.21	-0.66	-0.59	0.31
ACC							1.00	0.24	0.12	0.23
CFP									0.70	-0.22
TURN									1.00	-0.42
BAB										1.00

one-year period and short it if the prior one-year period return is negative. To capture the time-varying components of factor momentum, we also construct a conditional CH3 model (Cond-CH3) that incorporates investor sentiment into the model's β .⁴ Table 2 reports the annualized average returns of the long-leg, short-leg, and long-short strategies; Sharpe ratios, abnormal alphas (FF5- α , CH3- α , and Cond-CH3- α) of the long-short portfolios; and corresponding *t*-statistics. The sample period runs from January 2002 to December 2019.

The individual factor momentum is ubiquitous. Table 2 shows that eight factors produce significantly positive factor momentum returns that cannot be explained by the FF5 model. For the significant average returns, the long-short strategy earns from 8.05 % (t = 2.19) for the turnover rate to 18.15 % (t = 4.15) for the book-to-market ratio per year. The annualized Sharpe ratio peaks at 0.97 for the book-to-market ratio, showing economically sizable benefits. Among the factors that generate significant raw returns, the CH3 factor model explains firm size and illiquidity, as it suits the Chinese stock market better than others. The Conditional CH3 factor model produces smaller alphas than CH3. Moreover, the long legs play an important role in generating factor momentum returns, as they lead to higher values and *t*-statistics.

To summarize, the individual factors demonstrate economically sizable time-series momentum. Hence, factors' recent performance can forecast their future returns (Moskowitz et al., 2012; Gupta and Kelly, 2019).

3.2. Time-series factor momentum and cross-sectional factor momentum strategies

We next investigate the performance of the TSFM and CSFM strategies as they consider the 10 non-momentum factors in combination. Gupta and Kelly (2019) adopt the TSFM strategy, while Avramov et al. (2017) and Arnott et al. (2021) mainly investigate the CSFM strategy. Table 3 reports the annualized average returns, standard deviations, Sharpe ratios, abnormal alphas (FF5- α , CH3- α , and Cond-CH3- α), and corresponding *t*-statistics for the TSFM and CSFM strategies.

The TSFM strategy is similar to individual factor momentum but integrates the individual signals into a single portfolio. In each month, the TSFM strategy longs factors with positive performance over the prior one-year period and shorts those with negative performance. Panel A of Table 3 shows that the TSFM is a beneficial factor investment strategy, as it earns an annualized average return of 9.91 % (t = 4.88), with a Sharpe ratio of 1.15. Furthermore, the TSFM strategy delivers significant abnormal alphas, equaling 9.59 % (t = 7.03), 7.76 % (t = 3.35), and 7.04 % (t = 3.21) for FF5- α , CH3- α , and Cond-CH3- α , respectively. The long- and short-legs generate 14.07 % (t = 5.96) and 3.37 % (t = 0.97) returns per year and have Sharpe ratios of 1.40 and 0.23, respectively. The factor models cannot explain the returns on the long legs, although they subsume the short-leg returns well.

The CSFM strategy takes positions based on the factors' recent performance compared with the cross-sectional performance of all

⁴ The conditional CH3 factor model is: $\text{FFM}_t = \alpha + (1 + \text{Sentiment}_t)\beta_1 MKT_t + (1 + \text{Sentiment}_t)\beta_2 SIZE_t + (1 + \text{Sentiment}_t)\beta_3 BM_t + \varepsilon_t$, where Sentiment is Baker and Wurgler (2006)'s investor sentiment with the Chinese data.

Performance of Individual Factor Momentum.

This table reports the annualized average returns, Sharpe ratios (SR), abnormal alphas (FF5- α , CH3- α , and Cond-CH3- α), and corresponding *t*-statistics (in square brackets) for the individual factor momentum strategy (FM). For each month, we long the factor with positive returns over the prior one-year period and short it if the prior one-year period return is negative. We also report the average returns and *t*-statistics for the winner and loser portfolios. The portfolios are rebalanced monthly. The conditional CH3 model (Cond-CH3) imposes investor sentiment into the model's β . All the returns are reported in percentages. The sample period runs from January 2002 to December 2019.

	Winner	Loser	FM	SR	FF5-α	СНЗ-а	Cond-CH3-α
SIZE	21.36 [2.68]	-7.70 [-0.90]	15.99 [2.71]	0.63	8.51 [2.40]	3.83 [0.98]	3.56 [0.71]
BM	22.19 [4.49]	2.71 [0.35]	18.15 [4.15]	0.97	16.68 [4.92]	15.61 [3.17]	13.14 [3.04]
GP	14.18 [3.41]	3.84 [0.30]	10.60 [2.03]	0.58	11.04 [2.53]	12.91 [2.73]	10.25 [2.44]
CINVEST	-4.03 [-1.22]	-2.68 [-0.67]	-0.24 [-0.09]	-0.02	1.56 [0.41]	1.39 [0.34]	0.82 [0.20]
ILL	16.72 [2.84]	-3.97 [-0.47]	12.59 [2.60]	0.61	11.52 [2.70]	5.90 [1.33]	5.11 [1.12]
EP	13.66 [4.31]	24.17 [1.72]	10.45 [3.26]	0.76	12.68 [4.97]	-2.34 [-1.84]	-1.92 [-1.67]
ACC	11.24 [4.25]	14.91 [1.91]	9.43 [3.68]	0.86	11.76 [5.11]	8.52 [2.25]	8.25 [2.12]
CFP	11.19 [1.96]	-8.39 [-1.80]	9.90 [2.64]	0.62	6.96 [2.01]	9.29 [2.21]	8.54 [2.03]
TURN	12.98 [3.14]	11.18 [1.49]	8.05 [2.19]	0.51	5.70 [1.69]	11.32 [2.50]	10.44 [2.41]
BAB	14.00 [1.52]	4.53 [0.36]	7.65 [1.02]	0.24	9.40 [1.44]	9.84 [1.40]	7.21 [1.23]

Table 3

Performance of the TSFM and CSFM Strategies.

This table reports the annualized average returns, standard deviations, Sharpe ratios (SR), abnormal alphas (FF5- α , CH3- α , and Cond-CH3- α), and corresponding *t*-statistics (in square brackets) for the time-series factor momentum (TSFM) and cross-sectional factor momentum (CSFM) strategies using the 10 non-momentum factors. The TSFM strategy longs factors with positive returns over the prior year (winners) and shorts those with negative returns (losers). The CSFM strategy longs factors with above-median returns relative to other factors over the prior year (winners) and shorts those with below-median returns (losers). We rebalance all the strategies monthly. The conditional CH3 model (Cond-CH3) imposes the investor sentiment into the model's β . All the returns are reported in percentages. The sample period runs from January 2002 to December 2019.

	Mean	STD	SR	FF5-α	СНЗ-а	Cond-CH3-α
Panel A: Time-series f	actor momentum					
Full Sample	9.91 [4.88]	8.62	1.15	9.59 [7.03]	7.76 [3.35]	7.04 [3.21]
Winners	14.07 [5.96]	10.05	1.40	13.53 [8.55]	10.69 [4.07]	9.88 [3.75]
Losers	3.37 [0.97]	14.80	0.23	3.08 [1.27]	0.19 [0.06]	0.67 [0.42]
Panel B: Cross-section	al factor momentum					
Full Sample	7.02 [3.44]	8.78	0.80	7.20 [4.00]	6.10 [2.75]	5.72 [2.55]
Winners	16.95 [5.51]	13.04	1.30	16.00 [7.29]	13.59 [4.18]	11.24 [3.78]
Losers	2.91 [1.16]	10.64	0.27	2.60 [1.47]	1.40 [0.57]	1.17 [0.44]

factors. We long factors with above-median returns relative to other factors over the prior year and short those with below-median returns following Jegadeesh and Titman (1993). Panel B of Table 3 shows that the CSFM strategy produces 7.02 % (t = 3.44) average returns annually, with winners and losers generating 16.95 % (t = 5.51) and 2.91 % (t = 1.16) per annum, respectively. The Sharpe ratios reach 0.80, 1.30, and 0.27 for the full sample, long-legs, and short-legs, respectively. The FF5, CH3, and Cond-CH3 models produce significant abnormal returns for the full sample and long-legs but insignificant returns for the short-legs.

The TSFM and CSFM strategies share some commonalities. First, winner portfolios contribute more to overall returns than losers. Second, loser portfolios generate positive insignificant returns, which is similar to Ehsani and Linnainmaa's (2022) findings in the U.S. market. Third, the TSFM and CSFM strategies both produce higher Sharpe ratios than the individual strategies, thereby showing impressive economic benefits.

However, the two strategies are also different. The TSFM strategy generates higher returns and Sharpe ratios than the CSFM strategy. The standard deviations are also mostly lower for the TSFM strategy. These results are similar to those found in the United States (Ehsani and Linnainmaa, 2022) and globally (Gupta and Kelly, 2019). From a construction perspective, the TSFM method is a pure bet on the factor return autocorrelations, whereas the CSFM strategy is developed relative to a peer group and considers negative cross-serial covariance (Ehsani and Linnainmaa, 2022).

Fig. 1 plots the cumulative returns associated with the winner and loser portfolios in Table 3 as well as the equal-weighted portfolio of the 10 non-momentum factors. The winner TSFM strategy earns 222 % from January 2002 to December 2019, and the winner CSFM strategy earns 205 %. In contrast, both loser strategies generate cumulative returns close to zero. The equal-weighted portfolio of the 10 non-momentum factors generates a lower cumulative return by the end of the sample period, which reinforces factor momentum strategies' economic benefits.

For robustness, we extend our anomaly set by adding 50 more anomalies, (i.e., 60 anomalies in total) to investigate the performance of the factor momentum strategies, which covers value versus growth, investment, profitability, trading friction, and intangibles, based on the categories of Hou et al. (2020). The Online Appendix B presents the construction of the 50 anomalies. Table 4 reports the results. We find that factor momentum constructed based on 60 anomalies also has high average returns, Sharpe ratios, and abnormal alphas, as Table 4 shows. Panel A reports the results for the TSFM strategy based on the 60 anomalies. The average return is 6.18 % (t = 3.39),



Date

Fig. 1. Cumulative Returns

This figure plots the cumulative returns on factors divided into winners and losers based on their past performance and an equal-weighted portfolio of all 10 non-momentum factors from January 2002 to December 2019. Time-series winners and losers are factors with positive or negative returns over the past year. Cross-sectional winners and losers are factors that have above- or below-median returns over the prior year. The portfolios are rebalanced monthly, and each portfolio's standard deviation is standardized to equal that of the equal-weighted portfolio.

Table 4

Performance of the TSFM and CSFM Strategies with More Anomalies

This table reports the annualized average returns, standard deviations, Sharpe ratios (SR), abnormal alphas (FF5- α , CH3- α , and Cond-CH3- α), and corresponding *t*-statistics (in square brackets) for different combinations of 60 anomalies in China. The time-series factor momentum (TSFM) strategy longs factors with positive returns over the prior one-year period (winners) and shorts those with negative returns (losers). The cross-sectional factor momentum (CSFM) strategy longs factors with above-median returns relative to other factors over the prior year (winners) and shorts those with below-median returns (losers). We rebalance all the strategies monthly. The conditional CH3 model (Cond-CH3) imposes the investor sentiment into the model's β . All the returns are reported in percentages. The sample period runs from January 2002 to December 2019.

	Mean	STD	SR	FF5-α	СНЗ-а	Cond-CH3-α
Panel A: Time-series fac	ctor momentum					
Full Sample	6.18 [3.39]	7.73	0.80	6.96 [3.59]	4.92 [2.48]	4.89 [2.45]
Winners	7.10 [3.61]	8.35	0.85	7.92 [3.81]	5.04 [2.10]	4.97 [2.06]
Losers	-4.16 [-2.14]	8.24	-0.50	4.56 [2.24]	3.96 [1.77]	3.88 [1.97]
Panel B: Cross-sectional	l factor momentum					
Full Sample	5.81 [3.43]	7.19	0.81	6.36 [3.50]	4.56 [2.50]	4.51 [2.46]
Winners	8.71 [3.76]	9.83	0.89	9.60 [3.91]	6.00 [2.13]	5.88 [2.07]
Losers	-2.90 [-1.61]	7.65	-0.38	3.00 [1.53]	3.24 [1.62]	3.13 [1.55]

and the Sharpe ratio is 0.80. The abnormal alphas equal 6.96 % (t = 3.59), 4.92 % (t = 2.48), and 4.89 % (t = 2.45) for the FF5, CH3, and Cond-CH3 factor models. The CSFM strategy, reported in Panel B of Table 12, produces an average yearly return of 5.81 % (t = 3.43), a Sharpe ratio of 0.81, and positively significant abnormal alphas. In short, factor momentum is robust to the choice of the size of the anomaly set. Both the 10 factors and 60 anomalies yield substantial positive returns. However, for our main research focus, we adopt TSFM and CSFM with the 10 factors, as they generate higher profits compared to the 60 anomalies and are relatively more straightforward to construct (Ehsani and Linnainmaa, 2022; Li et al., 2022).

Overall, we find significant positive factor momentum in the Chinese stock market. The strategy earns higher returns in China than in the United States (Ehsani and Linnainmaa, 2022), but as the standard deviations of the returns are also larger, the Sharpe ratios of the two markets are similar.

3.3. Decomposing factor momentum profits

Following Lo and MacKinlay (1990), Lewellen (2002), and Ehsani and Linnainmaa (2022), we decompose the TSFM and CSFM strategies to clarify the sources of their profits. Here, we use the linear weight factor momentum strategy to run the Lo–MacKinlay decomposition (Ehsani and Linnainmaa, 2022).

For the TSFM strategy, the weight of factor f in month t is $w_t^f = r_{-t}^f$, where r_{-t}^f is factor f's past return over the formation period. According to Moskowitz et al. (2012), the expected return of the TSFM strategy can be written as

$$E\left[\pi_{t}^{TSFM}\right] = \frac{1}{F}E\left[\sum_{f=1}^{F} r_{-t}^{f} r_{t}^{f}\right] = \frac{1}{F}\sum_{f=1}^{F}\left[\operatorname{cov}(r_{-t}^{f}, r_{t}^{f}) + \left(\mu^{f}\right)^{2}\right],\tag{1}$$

where r_t^f is factor f's return in month t and μ^f is factor f's unconditional expected return. Eq. (1) suggests that the TSFM strategy can be decomposed into the factor return autocorrelations and mean squared returns.

As shown in Table 5, the TSFM strategy earns an annualized return of 17.83 % (t = 4.06). The factor returns autocorrelations generate an average return of 7.99 % (t = 2.00), and the mean-squared component delivers an average return of 9.84 % (t = 4.15) per annum. Hence, compared with Ehsani and Linnainmaa (2022), the mean-squared return contributes more in the Chinese stock market.

For the CSFM strategy, the weight of factor *f* in month *t* is $w_t^f = r_{-t}^f - \bar{r}_{-t}$, where \bar{r}_{-t} is the cross-sectional mean of all the factors' returns over the formation period. The expected return of the CSFM strategy can be stated as

$$E[\pi_{t}^{CSFM}] = \frac{1}{F}E\left[\sum_{f=1}^{F} \left(r_{-t}^{f} - \bar{r}_{-t}\right)r_{t}^{f}\right] = \frac{1}{F}\sum_{f=1}^{F}\left[\operatorname{cov}\left(r_{-t}^{f}, r_{t}^{f}\right) - \operatorname{cov}(\bar{r}_{-t}, \bar{r}_{t}) + \left(\mu^{f} - \bar{\mu}\right)^{2}\right],\tag{2}$$

where $\overline{\mu}$ is the cross-sectional average of all the factors' unconditional expected returns. Eq. (2) shows that the profits of the CSFM strategy can be separated by the factor return autocovariance, negative cross-serial covariance, and cross-sectional variance of mean returns.

Table 5 shows that the CSFM strategy earns an average annualized return of 9.92 % with a *t* value of 2.68. The autocovariance contributes an average annualized return of 5.22 % (t = 2.00). The cross-serial covariance term is positive and therefore makes a negative contribution, which is -1.04 % per annum. The positive cross-serial covariance means that a positive return on one factor accompanies positive returns on the other factors. However, as the CSFM strategy bets on the relative performance of a factor, this decreases its performance. The mean-squared component delivers an average return of 3.66 % (t = 3.24) per annum.

Of the two strategies, the TSFM strategy, which is a pure bet on the factor return autocorrelations, outperforms the CSFM strategy (Ehsani and Linnainmaa, 2022). Therefore, we use the TSFM portfolio for our further analyses.

3.4. Digest stock anomalies

Factor momentum is aggregated from individual anomalies, which motivates us to explore whether the combined factor momentum strategy explains anomalies in this subsection.

First, we investigate whether factor momentum can explain its components (i.e., the 10 non-momentum factors) in China. We regress each of the factor's returns on the CH3 factors and CH3 factors augmented with factor momentum:

$$FACTORRET_t = \alpha + \gamma CH3_t + \varepsilon_t,$$
(3)

$$FACTORRET_t = \alpha + \beta TSFM_t + \gamma CH3_t + \varepsilon_t,$$
(4)

where FACTORRET_t denotes each of the long-short spread portfolios of the 10 non-momentum factors at time t, TSFM_t is the return of

Table 5

Decomposition of Factor Momentum Profits

This table decomposes the profits of the time-series factor momentum (TSFM) and cross-sectional factor momentum (CSFM) profits using Eqs. (1) and (2). We report the annualized factor premiums in percentages and *t*-statistics in square brackets. When month *t* is sampled, we link month *t* with the factors' average returns from month t - 12 to t - 1 to compute the decomposition. The sample period runs from January 2002 to December 2019.

Decomposition	Mean	<i>t</i> -stat
Time-series factor momentum =	17.83	[4.06]
Autocovariances	7.99	[2.00]
+ Mean squared returns	9.84	[4.15]
Cross-sectional factor momentum =	9.92	[2.68]
Autocovariance	5.22	[2.00]
- Cross-serial covariances	1.04	[0.16]
+ Variance of mean returns	3.66	[3.24]

Digesting Stock Anomalies

Panel A compares the performance of the Chinese three-factor models (CH3) and CH3 augmented with the time-series factor momentum (TSFM) strategy in digesting the monthly excess returns on the 10 non-momentum factors. Panel B expands the test by explaining 50 anomalies. Panel C reports the average absolute values of all the regressions. We report the alphas and adjusted R^2s (in percentages) for both models, loadings of the TSFM factor, and differences between the two alphas (Diff). We report the alphas in percentages per year and the *t*-statistics in square brackets. The sample period runs from January 2002 to December 2019.

	CH3 Regression		TSFM+CH3 Regressio	on		Diff.
	СНЗ-а	R ² (%)	TSFM+CH3-α	TSFM- β	R ² (%)	
Panel A: Ten factors						
SIZE	0.84 [0.36]	89.67	0.96 [0.42]	0.23 [2.72]	90.25	0.12
BM	22.80 [4.75]	12.20	14.76 [3.55]	1.04 [6.13]	34.98	-8.04
GP	8.76 [2.42]	52.04	5.64 [1.64]	0.40 [2.86]	55.60	-3.12
CINVEST	0.48 [0.17]	11.01	0.36 [0.11]	0.02 [0.17]	11.03	-0.12
ILL	1.20 [0.68]	87.76	0.96 [0.52]	0.03 [0.47]	87.78	-0.24
EP	4.44 [2.12]	86.75	4.92 [2.34]	0.06 [0.79]	86.80	0.48
ACC	9.36 [2.54]	5.17	4.92 [1.50]	0.56 [5.03]	25.15	-4.44
CFP	4.32 [1.34]	52.10	0.96 [0.31]	0.43 [3.57]	57.20	-3.36
TURN	6.36 [1.86]	38.86	2.16 [0.65]	0.54 [5.26]	50.15	-4.20
BAB	0.72 [0.21]	60.06	0.96 [0.27]	0.03 [0.21]	60.07	0.24
Panel B: More Anomalie	s					
Value-versus-growth						
AM	19.44 [4.10]	12.86	11.40 [2.74]	1.05 [5.93]	35.36	-8.04
DA	6.00 [1.96]	41.06	1.56 [0.59]	0.57 [4.41]	50.28	-4.44
DER	14.76 [2.88]	6.60	5.88 [1.26]	1.14 [5.27]	29.83	-8.88
DLME	11.16 [2.99]	14.13	6.48 [1.86]	0.60 [3.74]	24.07	-4.68
LG	12.48 [4.60]	10.08	9.12 [3.53]	0.42 [4.04]	23.21	-3.36
SG	9.48 [2.78]	42.13	3.72 [1.21]	0.73 [5.50]	56.42	-5.76
SMI	4.32 [1.51]	2.20	1.92 [0.61]	0.30 [1.96]	7.15	-2.40
SP	14.76 [3.36]	23.69	7.08 [1.79]	1.00 [7.03]	47.44	-7.68
TG	6.00 [2.50]	8.71	2.88 [1.30]	0.40 [4.77]	23.09	-3.12
TOR	1.92 [0.90]	88.39	0.36 [0.18]	0.30 [2.88]	89.44	-1.56
Investment						
CAPXG	7.08 [3.00]	22.87	5.64 [2.29]	0.20 [2.33]	25.93	-1.44
IA	11.64 [3.45]	26.65	6.12 [1.87]	0.71 [5.08]	43.19	-5.52
IC	4.44 [1.69]	16.13	4.32 [1.67]	0.01 [0.09]	16.13	-0.12
IVC	7.68 [2.54]	14.68	3.60 [1.36]	0.52 [3.94]	28.92	-4.08
IVG	10.20 [3.60]	1.00	5.52 [2.03]	0.60 [4.32]	21.51	-4.68
Profitability						
CFOA	0.00 [0.01]	18.70	0.96 [0.30]	0.11 [0.93]	19.44	0.96
CHTX	0.84 [0.30]	70.49	0.36 [0.12]	0.16 [1.61]	71.19	-0.48
CTA	12.48 [2.98]	7.78	7.08 [1.91]	0.70 [4.48]	24.47	-5.40
CTO	5.76 [2.40]	33.93	5.52 [2.21]	0.03 [0.35]	33.99	-0.24
EPS	6.60 [2.57]	81.17	3.00 [1.26]	0.47 [5.65]	84.49	-3.60
EY	6.12 [2.71]	84.46	6.84 [3.15]	0.09 [1.08]	84.62	0.72
GM	9.24 [2.19]	18.04	2.28 [0.55]	0.90 [5.40]	40.22	-6.96
RNA	10.20 [4.04]	6.72	7.32 [3.07]	0.37 [5.02]	20.40	-2.88
ROA	12.48 [3.42]	54.76	6.00 [1.85]	0.84 [6.48]	67.55	-6.48
ROE	9.36 [2.84]	67.46	4.68 [1.58]	0.60 [5.17]	73.84	-4.68
ROIC	3.96 [1.50]	71.36	0.96 [0.43]	0.38 [4.78]	75.00	-3.00
SALEREV	0.36 [0.13]	76.69	0.24 [0.09]	0.08 [0.84]	76.80	-0.12
TBI	1.32 [0.56]	48.91	1.68 [0.73]	0.06 [0.69]	49.10	0.36
Z	4.20 [2.13]	84.02	3.00 [1.62]	0.17 [1.78]	84.65	-1.20
Trading friction						
BETA	5.16 [1.64]	74.33	5.04 [1.48]	0.01 [0.07]	74.33	-0.12
BETASQ	5.16 [1.64]	74.31	5.04 [1.48]	0.01 [0.08]	74.31	-0.12
BETA_DIM	0.48 [0.16]	43.40	1.08 [0.36]	0.08 [0.60]	43.60	0.60
BETA_DN	2.28 [0.70]	61.06	2.16 [0.58]	0.03 [0.18]	61.07	-0.12
BETA_FF	1.56 [0.50]	55.80	1.80 [0.54]	0.03 [0.22]	55.81	0.24
BETA_HS	3.48 [0.92]	58.05	3.36 [0.84]	0.03 [0.19]	58.06	-0.12
IVOL	5.76 [1.58]	27.76	3.48 [0.92]	0.29 [1.35]	29.66	-2.28
MAXRET	18.60 [3.37]	25.15	18.72 [3.24]	0.00 [0.03]	25.15	0.12
PRC	10.68 [2.16]	9.72	2.64 [0.56]	1.03 [4.63]	27.80	-8.04
PRCDEL	0.24 [0.10]	11.51	1.44 [0.57]	0.15 [1.31]	13.33	1.20
RETVOL	15.84 [3.59]	59.02	14.28 [3.08]	0.21 [1.48]	59.64	-1.56
RVOL	7.20 [2.64]	77.12	8.40 [3.23]	0.15 [1.68]	77.57	1.20
STD_RVOL	6.36 [1.38]	3.78	6.96 [1.42]	0.07 [0.30]	3.92	0.60
STD_TURN	7.32 [2.12]	38.14	3.00 [0.91]	0.55 [4.92]	48.56	-4.32

(continued on next page)

Table 6 (continued)

	CH3 Regression		TSFM+CH3 Regression	Diff.		
	СНЗ-а	R ² (%)	TSFM+CH3-α	TSFM- β	R ² (%)	
VOLT	8.52 [2.19]	4.72	6.00 [1.52]	0.33 [2.39]	8.44	-2.52
ZEROTRADE	7.80 [2.22]	34.81	3.48 [1.02]	0.55 [5.02]	46.18	-4.32
Intangibles						
CFD	11.88 [3.00]	40.02	4.68 [1.33]	0.93 [5.96]	56.59	-7.20
CP	15.96 [3.16]	16.49	8.28 [1.85]	0.99 [4.88]	35.46	-7.68
CRG	2.28 [1.12]	29.76	1.92 [0.97]	0.03 [0.43]	29.87	-0.36
SC	6.24 [1.70]	20.04	1.32 [0.39]	0.64 [5.52]	37.02	-4.92
SI	5.88 [1.65]	22.57	3.36 [0.98]	0.33 [2.53]	27.06	-2.52
Panel C: Average Abs	solute Value					
Average	7.20 [2.06]	38.98	4.48 [1.36]	0.39 [2.94]	46.34	-2.72

the TSFM strategy, and $CH3_t$ is the CH3 model.

Panel A of Table 6 compares the performance of CH3 and CH3 augmented with the TSFM strategy to digest the excess returns on the 10 non-momentum factors, including the alphas and adjusted R^2 s for both models, loadings of the TSFM strategy, and differences between the TSFM+CH3 and CH3 alphas.

We find that controlling for the TSFM decreases the abnormal alphas and corresponding *t*-statistics in seven of the 10 regressions. For example, the CH3 alpha for book-to-market equity is 22.80 % (t = 4.75), whereas its TSFM+CH3 alpha reduces to 14.76 % (t = 3.55). The loadings of the TSFM strategy are mostly positively significant. Furthermore, the R^2 s increase in all 10 regressions. The results suggest the additional explanatory power of the TSFM strategy to subsume the factors compared with the CH3 factors.

Table 7

Stock Momentum and Factor Momentum

Panel A reports the annualized average returns, standard deviation, and annualized Sharpe ratios (SR) of the momentum strategies, i.e., stock momentum, high-priced stock momentum, industry momentum, and reversal, respectively. Stock momentum is calculated as the difference between the average value-weighted returns in the top quintiles and those in the bottom quintiles sorted by the stocks prior to the 12- to 2-month returns. High-priced stock momentum is the momentum in the highest 10 % price deciles; specifically, we independently sort stocks into deciles based on nominal prices and quintiles based on cumulative returns over the past 12 months. Industry momentum portfolio longs the top 20 % of industry portfolios that earn the largest returns in the prior k to 2 months and shorts the bottom 20 %. Reversal is constructed as the difference between the average value-weighted returns in the top quintiles and that in the bottom quintiles ranked by the stocks prior to one-month returns.

Panel B reports the results of regressing the return of the time-series factor momentum (TSFM) strategy on each of the four momentum strategies and FF5 separately:

 $Ret_t^{TSFM} = \alpha + \beta Ret_t^M + \gamma FF5_t + \varepsilon_t, \ M \in \{MOMe, MOMe_{Prc}, MOMe_{Ind}, REV\}.$

Panel C reports the results of regressing each of the returns on the four momentum strategies on the return of the TSFM and FF5 separately:

 $Ret_t^M = \alpha + \beta Ret_t^{TSFM} + \gamma FF5_t + \varepsilon_t, M \in \{MOMe, MOMe_{Prc}, MOMe_{Ind}, REV\}.$

The intercepts and returns are reported in percentages. We report the adjusted R^2 s (in percentages). The *t*-statistics are reported in square brackets. The sample period runs from January 2002 to December 2019.

Panel A			
	MEAN	STD	SR
МОМе	6.00 [1.03]	24.71	0.24
MOMe _{Prc}	12.31 [2.59]	20.76	0.60
MOMe _{Ind}	9.03 [2.42]	15.40	0.59
REV	-13.46 [-2.40]	23.76	-0.57
Panel B			
	Intercept	Slope (Ret_t^M)	R ² (%)
МОМе	5.73 [3.21]	0.20 [7.41]	40.34
MOMeprc	4.56 [2.05]	0.16 [6.31]	24.00
MOMe _{Ind}	6.83 [3.29]	0.32 [4.11]	22.79
REV	7.98 [3.80]	0.09 [2.58]	14.80
Panel C			
	Intercept	Slope (Ret_t^{TSFM})	R ² (%)
МОМе	-4.29 [-0.80]	1.69 [8.59]	39.22
MOMe _{Prc}	1.20 [0.51]	1.00 [5.86]	20.30
MOMe _{Ind}	0.71 [0.29]	0.38 [4.25]	13.56
REV	-11.35 [-2.07]	0.64 [2.85]	20.81

Then, we expand the investigation by including 50 anomalies in the Chinese stock market, as discussed in Section III. A (i.e., 60 anomalies in total). We regress each of the anomaly's returns on the CH3 factors and factor momentum portfolio:

$$ANOMALYRET_t = \alpha + \gamma CH3_t + \varepsilon_t, \tag{5}$$

ANOMALYRET_t =
$$\alpha + \beta TSFM_t + \gamma CH3_t + \varepsilon_t$$
, (6)

where ANOMALYRET_t denotes each of the long-short spread portfolios of the 50 anomalies at time t, TSFM_t is the return of the TSFM strategy of the 10 non-momentum factors, and CH3_t is the CH3 model.

Panel B of Table 6 reports that the TSFM strategy shows additional power in explaining the 50 anomalies. Accounting for the TSFM reduces the abnormal alphas and *t*-statistics in all the value versus growth group anomalies, all the investment-related anomalies, 11 of the 14 profitability-related characteristics, 10 of the 16 trading friction anomalies, and all the intangibles. In other words, 82 % of the anomalies produce lower abnormal alphas when controlling for the TSFM. Furthermore, 16 anomalies have insignificant TSFM+CH3 alphas despite producing significant CH3 alphas, while 62 % of the anomalies produce significantly positive TSFM coefficients. Adding the TSFM as a control variable increases the R^2 s, representing the goodness of fit, of 47 regressions.

Additionally, we report the average absolute values of 60 anomalies. Controlling for the TSFM makes the significant 7.20 % intercept (t = 2.06) drop 2.72 percentage points to become insignificant at 4.48 % (t = 1.36). The beta loading of factor momentum is 0.39 (t = 2.94), and R^2 reaches 46.34 %. These average absolute values show that factor momentum helps explain anomalies.

Our results illustrate that the TSFM portfolio helps decompose the returns on its component factors and a higher number of anomalies. Only a few factors aligned explain nearly 82 % of the anomalies, which indicates the strong economic potential of factor momentum. Dong et al. (2022) use several shrinkage approaches to extract predictive signals from 100 anomalies and find that long-short anomaly portfolio returns are statistically and economically significant in forecasting market excess returns. Our findings also shed light on the economic linkages between time-series factors and cross-sectional anomalies.

4. Factor momentum vis-à-vis stock momentum

4.1. Regressions of factor momentum and stock momentum

Previous studies have discussed the relation between factor momentum and stock momentum. For example, Arnott et al. (2021) highlight that industry momentum comes from factor momentum. Leippold and Yang (2021) argue that stock momentum is a combination of factor momentum. In this subsection, we investigate the relation between factor momentum and stock momentum in China, in the spirit of Arnott et al. (2021), Ehsani and Linnainmaa (2022), and Leippold and Yang (2021).

The Chinese stock market lacks momentum on aggregate (e.g., Pan et al., 2013; Boubaker et al., 2021). We calculate stock-level momentum (MOMe) as the difference between the top quintile's average value-weighted return and the lowest quintile's, ranked by stocks' prior 12- to 2-month returns. Panel A of Table 7 shows that the stock-level momentum in China is insignificant, with an annualized return of 6.00 % and a *t*-statistic of 1.03, in line with the prior literature.

Many studies consider different subsamples and high-frequency data, finding intraday (overnight; Gao et al., 2021), idiosyncratic (Lin, 2022), industry (Boubaker et al., 2021), and down-market stock momentums (Cheema and Nartea, 2017) in China. Specifically, Du et al. (2022) show that momentum in high-priced stocks is significant and document that retail investors' noise trade attenuates momentum, for small retail investors in China are constrained to invest in high-priced stocks because of the minimum 100 unit of the number of shares. Following Du et al. (2022), we calculate the high-priced stock momentum strategy (MOMe_{Prc}). We sort stocks into deciles based on their previous month's nominal price. Independently, we sort stocks into quintiles based on their prior 12 months' cumulative returns. The high-priced momentum strategy longs in past winner stocks and shorts the past losers in the highest price decile. We rebalanced the portfolios monthly and report the equal-weighted monthly returns. Panel A of Table 7 reports a significant high-priced stock momentum in China. The annualized return equals 12.31 % (t = 2.59) with a Sharpe ratio of 0.60.⁵

We also test several types of individual stock momentum in China following Ehsani and Linnainmaa (2022): standard (Jegadeesh and Titman, 1993), industry-adjusted (Cohen and Polk, 1998), industry (Moskowitz and Grinblatt, 1999), intermediate (Novy-Marx, 2012), and Sharpe ratio momentums (Rachev et al., 2007). Of these, only the industry momentum approach is significant, consistent with Boubaker et al. (2021). In detail, we classify stocks into 90 industries based on the Guidelines for the Industry Classification of Listed Companies of the China Securities Regulatory Commission (CSRC). The industry momentum portfolio (MOMe_{Ind}) longs the top 20 % of industry portfolios that earn the largest returns in the prior 12 to 2 months and shorts the bottom 20 % (Asness et al., 2000; Leippold and Yang, 2021; Moskowitz and Grinblatt, 1999). We rebalance the portfolios monthly. The results of Panel A of Table 7 show that the Chinese stock market delivers positive industry momentum, with an average return of 9.03 % (t = 2.42) and a Sharpe ratio of 0.59 per annum.

Finally, we calculate the reversal factor (REV) as that ranked by the previous one-month returns. Reversal is significantly negative in China, generating a -13.46 % return per year with a *t* value of -2.40. The high participation of short-term trade retail investors (Jones et al., 2020) leads to significant short-term reversals (Du et al., 2022).

⁵ We calculate the performance of momentum and time-series factor momentum across priced-based deciles and report the results in Online Appendix Table OA1. The results show that momentum is significant only in the top price decile and TSFM remains significant in nine decile groups.

To examine the ability of the four momentum (reversal) strategies to explain the factor momentum, we run the following regression:

$$\operatorname{RET}_{t}^{TSFM} = \alpha + \beta \operatorname{RET}_{t}^{M} + \gamma \operatorname{FF5}_{t} + \varepsilon_{t}, \ M \in \{\operatorname{MOMe}, \operatorname{MOMe}_{\operatorname{Prc}}, \operatorname{MOMe}_{\operatorname{Ind}}, \operatorname{REV}\},$$
(7)

where $\text{RET}_t^{\text{TSFM}}$ is the return of the TSFM strategy, RET_t^M is the return of each of the stock momentum strategies, $M \in \{\text{MOMe}, \text{MOMe}_{\text{Prc}}, \text{MOMe}_{\text{Ind}}, \text{REV}\}$, and FF5_t is the FF5 model. Panel B of Table 7 shows the results.

The findings suggest that the four momentum (reversal) strategies cannot subsume factor momentum returns, as shown in the intercepts. Factor momentum generates significant positive alphas of 5.73 % (t = 3.21), 4.56 % (t = 2.05), 6.83 % (t = 3.29), and 7.98 % (t = 3.80) after adjusting for MOMe,MOMe_{Prc},MOMe_{Ind},and REV, respectively. The loadings on the momentum (reversal) portfolios are all positively significant, indicating that they partially explain the factor momentum. R^2 reaches the highest value of 40.34 % for the momentum portfolio.

Next, we investigate whether the TSFM strategy helps explain the returns of the four momentum (reversal) portfolios, with the following regression:

$$\operatorname{RET}_{t}^{M} = \alpha + \beta \operatorname{RET}_{t}^{\text{TSFM}} + \gamma \operatorname{FF5}_{t} + \varepsilon_{t}, \ M \in \{\operatorname{MOMe}, \operatorname{MOMe}_{\operatorname{Prc}}, \operatorname{MOMe}_{\operatorname{Ind}}, \operatorname{REV}\}.$$
(8)

Panel C of Table 7 reports the results of Eq. (8). We find that the TSFM subsumes the return of the three stock momentum strategies, with an insignificant alpha of -4.29 % (t = -0.80), 1.20 % (t = 0.51), and 0.71 % (t = 0.29) of MOMe, MOMe_{Prc}, and MOMe_{Ind}, respectively. However, the reversal portfolio remains negatively significant when controlling for factor momentum, with an abnormal alpha of -11.35 % (t = -2.07). The coefficients on the factor momentum are all positively significant.

In sum, we find that factor momentum subsumes momentum, high-priced momentum, and industry momentum in China. As Ehsani and Linnainmaa (2022) employ momentum-neutral factors and find that factor momentum exists in momentum-neutral factors, our findings indicate that factor momentum is distinct from stock momentum.

4.2. Market timing of factor momentum and stock momentum

Literature suggests that price momentum exhibits negative returns during market downturns, commonly referred to as momentum crashes (e.g., Daniel and Moskowitz, 2016; Dierkes and Krupski, 2022). We investigate whether market conditions affect the performance of factor momentum and stock momentum following Daniel and Moskowitz (2016) with the three regressions:

$$\operatorname{RET}_{t}^{\kappa} = \alpha_{0} + \beta_{0} R_{m,t} + \varepsilon_{t}, \tag{9}$$

$$\operatorname{RET}_{t}^{K} = \left(\alpha_{0} + \alpha_{B}I_{B,t-1}\right) + \left(\beta_{0} + \beta_{B}I_{B,t-1}\right)R_{m,t} + \varepsilon_{t},\tag{10}$$

$$\operatorname{RET}_{t}^{K} = \left(\alpha_{0} + \alpha_{B}I_{B,t-1}\right) + \left(\beta_{0} + I_{B,t-1}\left(\beta_{B} + I_{U,t}\beta_{B,U}\right)\right)R_{m,t} + \varepsilon_{t}, K \in TSFM, MOMe,$$

$$(11)$$

where RET_{t}^{TSFM} represents the returns of the time-series factor momentum, RET_{t}^{MOMe} is the zero-investment winner-minus-loser portfolio, which longs the top decile and shorts the bottom decile based on cumulative returns from t - 12 to t - 2-month, and $R_{m,t}$ is the value-weighted market excess returns in month t. The indicator $I_{B,t-1}$ denotes the bear market state, which equals one if the cumulative market return is negative over the past 24 months and zero otherwise, while $I_{U,t}$ stands for the bull market state, which is one if the excess market return is greater than the risk-free rate in month t and zero otherwise.

Panel A of Table 8 reports the results of TSFM for Eqs. (9)-(11) in columns (1)-(3), respectively. The findings indicate that the factor momentum returns are robust for the most part and exhibit asymmetry only after market downturns. Column (1) reports a negative estimated market beta of -0.031 with an insignificant *t*-statistic of -1.14, and an intercept of 0.87 (t = 5.33), indicating that the market return does not affect factor momentum. We then introduce a dummy variable for the bear market state and an interaction term in Eq. (10) to capture the differences in expected returns and market betas during bear markets. We observe a significant negative $\hat{\beta}_B$ of -0.12 (t = -2.27), suggesting that factor momentum returns decrease following a bear market period. In column (3), we include a contemporaneous up-market indicator to examine factor momentum returns under up- and down-market conditions. The insignificant $\hat{\beta}_{B,U}$ alleviates the concern that factor momentum is influenced when the market rebounds following a bear market, although stock momentum in the U.S. is affected (Daniel and Moskowitz, 2016). Nevertheless, Panel B of Table 8 presents the results of stock-level momentum, which shows that the returns of stock momentum exhibit asymmetry following bear markets.

Overall, our findings indicate that factor momentum returns are mostly robust and show asymmetries after downturn markets.

5. Economic channels of factor momentum

In this section, we aim to explore the economic mechanisms underlying factor momentum. First, we examine whether the mispricing correction brought by slow-moving arbitrage capital and limits-to-arbitrage is the source of factor momentum (Li et al., 2022). Instead of immediately eliminating all mispricing, arbitragers who seek to generate sustained factor returns attempt to correct the mispricing gradually through trading. Thus, during periods of high limits-to-arbitrage, arbitrageurs are more inclined to participate in trading to decrease synchronization risk (Abreu and Brunnermeier, 2002). We hypothesize that factor momentum is more pronounced during high limits-to-arbitrage periods. Besides, it takes time to make the price return to its fundamental value. Second, we decompose

Market Timing Regression Results

Panels A and B of this table report the results of three monthly time-series regressions specifications of factor momentum and stock momentum, respectively. The dependent variable is the return on the TSFM for Panel A and the return on the winner-minus-loser stock momentum portfolio for Panel B. The independent variables include a constant (1); a dummy variable representing bear markets ($I_{B,t-1}$), which equals one if the market's cumulative past two-year return is negative; the excess market return (R_{mkt}); and an indicator for up-market ($I_{U,t}$), which equals one if R_{mkt} is positive. The sample period runs from January 2002 to December 2019.

Coefficient	Variable	Estimated Coefficients		
		(1)	(2)	(3)
Panel A Factor Momentum				
$\widehat{\alpha}_0$	1	0.87 [5.33]	0.80 [3.25]	0.80 [3.26]
$\widehat{\alpha}_B$	$I_{B,t-1}$		0.17 [0.51]	0.01 [1.41]
$\widehat{\beta}_0$	R _{mkt}	-0.03 [-1.14]	0.001 [0.03]	0.001 [0.03]
$\hat{\beta}_B$	$I_{B,t-1} \cdot R_{mkt}$		-0.12 [-2.27]	-0.02 [-0.27]
$\hat{\beta}_{B,U}$	$I_{B,t-1} \cdot I_{U,t} \cdot R_{mkt}$			-0.18 [-1.34]
Panel B Stock Momentum				
$\widehat{\alpha}_0$	1	0.67 [1.54]	-0.40 [-0.61]	-0.40 [-0.60]
$\widehat{\alpha}_B$	$I_{B,t-1}$		0.02 [2.56]	0.03 [2.90]
$\widehat{\beta}_{0}$	R _{mkt}	-0.03 [-0.57]	0.04 [0.65]	0.04 [0.66]
$\hat{\beta}_B$	$I_{B,t-1} \cdot R_{mkt}$		-0.28 [-2.55]	-0.32 [-1.14]
$\hat{eta}_{B,U}$	$I_{B,t-1} \cdot I_{U,t} \cdot R_{mkt}$			-0.10 [-0.54]

the factor momentum strategy into buy-and-hold portfolios and factor time (Leippold and Yang, 2021) to analyze the contribution of each component. Third, we examine the impact of culture on factor momentum, as several studies have shown its influence on stock momentum (Chui et al., 2010).

5.1. Slowing-Moving mispricing correction

5.1.1. Aggregate idiosyncratic volatility

Frictions such as limited risk-bearing capacity and transaction costs cause arbitrageurs to respond slowly to mispricing, resulting in the persistence of asymmetric mispricing correction (Gârleanu and Pedersen, 2013, 2016; Dong et al., 2020; Dong et al., 2022). We suppose that factor momentum performs better during times of high aggregate idiosyncratic volatility (Li et al., 2022), as arbitrageurs are more likely to take advantage of high information asymmetry. Furthermore, high aggregate idiosyncratic volatility may cause large limits to arbitrage, leading to a scarcity of funds available for arbitrageurs and thus a long response time to correct mispricing.

Following Garcia et al. (2014) and Li et al. (2022), we calculate the monthly return dispersion to estimate aggregate idiosyncratic volatility, which is the cross-sectional standard deviation of daily stock returns. We calculate the average annualized returns of the factor momentum, winner, and loser portfolios during high and low return dispersion periods, following Stambaugh et al. (2012), according to whether they are above or below the mean of return dispersion, respectively.

Table 9 reports the results for the factor momentum strategies during periods of high and low aggregate idiosyncratic volatility.

Table 9

The Role of Idiosyncratic Volatility

Panels A and B report the annualized average returns and corresponding *t*-statistics (in square brackets) for the time-series factor momentum (TSFM) strategy, winners, and losers for stocks in low and high aggregate idiosyncratic volatility periods, respectively. The average monthly returns in low and high aggregate idiosyncratic volatility periods are estimates of a_L and a_H in the regression

$$R_{i,t} = \alpha_H d_{H,t} + \alpha_L d_{L,t} + \epsilon_{i,t},$$

where $d_{H,t}$ and $d_{L,t}$ are dummy variables indicating high- and low-aggregate idiosyncratic volatility periods. Panel C reports the difference in annualized average returns between the high and low idiosyncratic volatility periods. We calculate the monthly return dispersion to estimate aggregate idiosyncratic volatility, which is the cross-sectional standard deviation of daily stock returns. The high (low) aggregate idiosyncratic volatility period corresponds to the sample period with the above (below) mean return dispersion level. All the returns are reported in percentages. The sample period runs from January 2002 to December 2019.

	TSFM	Winners	Losers				
Panel A: High aggregate idiosynci	Panel A: High aggregate idiosyncratic volatility						
Mean	4.74	9.03	9.61				
t-stat	[2.02]	[3.09]	[2.24]				
Panel B: Low aggregate idiosyncr	atic volatility						
Mean	2.14	0.93	-6.45				
t-stat	[0.98]	[0.38]	[-1.81]				
Panel C: Difference between high and low aggregate idiosyncratic volatility							
Return	2.60 [0.61]	8.10 [1.59]	16.06 [2.14]				

Consistent with our conjecture, the factor momentum effect is stronger under high aggregate idiosyncratic volatility. The factor momentum strategy delivers an annualized return of 4.74 % (t = 2.02) during the high aggregate idiosyncratic volatility period compared with 2.14 % (t = 0.98) under low aggregate idiosyncratic volatility. The returns on the long- and short-legs reach higher values during high return dispersion periods (9.03 with 3.09 *t*-statistic and 9.61 with 2.24 *t*-statistic, respectively), whereas these values are lower (0.93 with 0.38 *t*-statistic and -6.45 with -1.81 *t*-statistic, respectively) during low return dispersion periods. In summary, the findings show stronger factor momentum during periods with increased aggregate idiosyncratic volatility.

5.1.2. Investor sentiment

Furthermore, we investigate the role of investor sentiment in factor momentum performance. According to Stambaugh et al. (2012), many anomalies generate high returns during high investor sentiment periods, and are particularly persistent in the presence of short-sale constraints because of mispricing. Factor momentum may be a different manifestation of the same mechanism, as it takes time to correct mispricing to the fundamental level due to short-sale constraints and slow-moving capital (Duffie, 2010). Thus, we presume that factor momentum is stronger during low investor sentiment periods than during high periods.

To test our hypothesis, we employ Baker and Wurgler's (2006) investor sentiment index in China and calculate the average annualized returns of the winner-minus-loser, winner, and loser portfolios during the high (above the mean) and low (below the mean) investor sentiment periods. Table 10 reports the results.

During the low investor sentiment period, the average underperforming factor earns a return of -2.97 % (t = -0.88), whereas factors with positive returns over the prior year earn a premium of 3.70 % (t = 1.59); this 4.10 % difference is significant with a t-statistic of 1.92. The average returns on the long-leg (5.94 % with 2.00 t-statistic) and short-leg (5.58 % with 1.26 t-statistic) are higher during the high investor sentiment period, but factor momentum strategy earns a lower benefit (2.64 % with 1.10 t-statistic). These findings are consistent with Stambaugh et al. (2012) study, which indicates that the average factor delivers high returns during high investor sentiment periods. However, the factor momentum strategy, as measured by the winner minus loser portfolio, generates lower returns during high sentiment periods. As it takes time for mispricing to develop, assets' values do not adjust immediately to the fundamental value due to limits to arbitrage. Factor momentum may therefore result from assets' values drifting back to their fundamental values.

5.1.3. Information asymmetry

The literature documents that information asymmetry influences cross-sectional stock momentum (Chen and Zhao, 2012; Gutierrez and Kelley, 2008). According to Zhang (2006), stocks with higher information uncertainty generate larger cross-sectional stock momentum because this intensifies the higher expected returns following good news and lower expected returns following bad news. Hence, we conjecture that information asymmetry impacts the factor momentum strategy.

Following Roll (1988), Jin and Myers (2006), and Boubaker et al. (2014), we exploit stock price nonsynchronicity (NONSYNC) as a measure of stock price informativeness. The higher NONSYNC is, the more informativeness and the less information asymmetry of stocks. We conjecture that stocks of higher (lower) factor momentum are less (more) informative and have a lower (higher) value of NONSYNC. To construct NONSYNC, we first run the following regression:

$$R_{i,t} = \alpha + \beta_1 R_{m,t} + \beta_2 R_{m,t-1} + \beta_3 R_{ind,t} + \beta_4 R_{ind,t-1} + \epsilon_{i,t}$$
(12)

where $R_{i,t}$ is the return on stock *I* on day *t*, R_m is the value-weighted market return, and R_{ind} is the value-weighted industry return. The regression is estimated for each firm *i* in a year. Then, we use the R_i^2 of the above regression to construct NONSYNC_i in a given year.

Table 10

The Role of Investor Sentiment

Panels A and B report the annualized average returns and corresponding *t*-statistics (in square brackets) for the time-series factor momentum (TSFM) strategy, winners, and losers for stocks in low and high investor sentiment periods, respectively. We reconstruct Baker & Wurgler's (2006) investor sentiment index in the Chinese market. The average monthly returns in low and high sentiment periods are estimates of α_L and α_H in the regression

 $R_{i,t} = \alpha_H d_{H,t} + \alpha_L d_{L,t} + \epsilon_{i,t}$

where $d_{H,t}$ and $d_{L,t}$ are dummy variables indicating high and low sentiment periods. Panel C reports the difference in annualized average returns between the high and low investor sentiment periods. The high (low) investor sentiment period corresponds to the sample period with the above (below) mean. All the returns are reported in percentages. The sample period runs from January 2002 to December 2019.

	TSFM	Winners	Losers			
Panel A: High sentiment						
Mean	2.64	5.94	5.58			
t-stat	[1.10]	[2.00]	[1.26]			
Panel B: Low sentiment						
Mean	4.10	3.70	-2.97			
t-stat	[1.92]	[1.59]	[-0.88]			
Panel C: Difference between high and low sentiment						
Return	-1.46 [-0.33]	2.24 [0.44]	8.55 [1.14]			

Following Morck et al. (2000), we logistically transform $1 - R_i^2$ to obtain an appropriate distribution, NONSYNC_i = ln $\left(\frac{1-R_i^2}{R^2}\right)$.

We then sort the stocks into two groups using NONSYNC at the beginning of each year and calculate the performance of the TSFM strategy for each group. Panels A and B of Table 11 report the annualized average returns, Sharpe ratios, and *t*-statistics for the full sample and long-leg and short-leg portfolios for stocks that have low and high information asymmetry, respectively. Consistent with our conjecture, factor momentum performance is high among stocks with severe information asymmetry. The TSFM strategy delivers an annualized return of 7.46 % (t = 2.99) among stocks with high information asymmetry, while the return is 3.19 % with a *t*-statistic of 0.84 among stocks with low information asymmetry. Both winners and losers with high information asymmetry expand with higher absolute return values. The Sharpe ratios are also higher for stocks with high information asymmetry in the long-short, long-leg, and short-leg samples. To summarize, factor momentum is higher if information asymmetry is relatively high.

5.1.4. Short-Sale constraints

To assess whether factor momentum is sensitive to short-sale constraints, we consider the factor momentum return in short-selling stocks and stocks with short-sale constraints. Restrictions on short-selling in the Chinese stock market are severe. In March 2010, the CSRC introduced the securities margin trading program in which only certain brokerage companies can short-sale 90 stocks on a special list. The program was expanded to 1600 stocks in 2019. However, short-sale constraints are still binding in China. The literature shows that short-selling helps price discovery, which increases market efficiency (Leippold et al., 2021). Therefore, we consider the role of short-sale constraints in factor momentum using data on short-sale stocks obtained from the CSRC. We separate stocks into two subsamples: one that allows short selling and the other with short selling constraints. Then, in each group, we run the TSFM strategy and report the results in Table 12.

The factor momentum strategy produces higher long-short returns among stocks without short-sale constraints, reaching 6.78 % (t = 3.87) per annum, while stocks under short-selling constraints have an average return of 6.54 % (t = 1.82). The difference in the average return between the two subsamples is 0.24 percentage points, with an insignificant *t*-statistic of 0.20. Both the long- and short-legs of stocks without short-sale constraints generate higher returns than stocks with short-sale constraints. The Sharpe ratio is also higher among non-short-selling stocks (1.36) compared with 0.64 for short-selling stocks.

Miller (1977) documents that short-selling tends to temper the propensity for riskier stocks to be bid up to higher prices. In other words, when short-selling is hard to execute, such as in the Chinese stock market, stocks tend to be overvalued relative to their fundamentals. Hong and Stein (2003) argue that the removal of short-selling constraints can incorporate unrevealed negative information into stock prices in a timely manner. Our results are consistent with theirs; the factor momentum strategy earns lower returns on short-selling stocks, which are likely to be priced more rationally with fundamental information, than stocks with short-sale constraints. The analysis is also consistent with the results in Table 12, as non-short-selling stocks also suffer higher information asymmetry.

Overall, our findings suggest that the slow-moving mispricing correction and the higher participation of arbitrageurs during the high limits-to-arbitrage periods help explain the returns of factor momentum.

5.2. Factor timing and factor premiums

Factor momentum combines factor timing and factor premiums. In this subsection, we decompose the factor momentum strategy into the factor timing (FT) and buy-and-hold (BH) portfolios following Leippold and Yang (2021) to investigate which portfolio contributes more. Timing may improve investment performance and capture time-varying risk, especially in a market with high volatility, such as China (Leippold et al., 2021). Furthermore, exposure to factors also leads to a risk premium.

The factor timing portfolio isolates the advantages of factor timing from those of factor premiums. If previous returns have deviated from the prevailing average return, it takes the corresponding factor positions; it longs factors when their returns are above their

Table 11

The Role of Information Asymmetry

Panels A and B report the annualized average returns, Sharpe ratios (SR), and corresponding *t*-statistics (in square brackets) for the time-series factor momentum (TSFM) strategy, winners, and losers for stocks with low and high information asymmetry, respectively. Panel C reports the difference in annualized average returns between high and low information asymmetry stocks. We exploit stock price nonsynchronicity (NONSYNC) as a measure of stock price informativeness. The higher NONSYNC is, the less information asymmetry stocks. The high (low) information asymmetry stocks correspond to the sample with the above (below) mean of NONSYNC. All the returns are reported in percentages. The sample period runs from January 2002 to December 2019.

	TSFM	Winners	Losers					
Panel A: Low information	Panel A: Low information asymmetry							
Mean	3.19 [0.84]	3.79 [1.03]	-2.98 [-0.61]					
SR	0.35	1.17	-0.24					
Panel B: High information	asymmetry							
Mean	7.46 [2.99]	5.04 [1.37]	-7.05 [-1.80]					
SR	1.20	1.68	0.76					
Panel C: Difference between high and low information asymmetry								
Return	4.27 [1.26]	1.25 [0.27]	-4.07 [-0.95]					

The Role of Short-Sale Constraints

Panels A and B report the annualized average returns, Sharpe ratios (SR), and corresponding *t*-statistics (in square brackets) for the time-series factor momentum (TSFM) strategy, winners, and losers for stocks allowed to engage in short selling (Short) and stocks prohibited from engaging in short selling (Nonshort), respectively. Panel C reports the difference in annualized average returns between stocks allowed to engage in short selling and those that cannot. All the returns are reported in percentages. The sample period runs from January 2010 to December 2019.

	TSFM	Winners	Losers		
Panel A: Short					
Mean	6.54 [1.82]	7.84 [2.36]	-4.48 [-0.90]		
SR	0.64	1.17	-0.24		
Panel B: Nonshort					
Mean	6.78 [3.87]	4.95 [2.13]	-8.52 [-3.40]		
SR	1.36	1.68	0.76		
Panel C: Difference between nonshort and short					
Return	0.24 [0.20]	-2.89 [-0.68]	-4.04 [-0.77]		

prevailing mean and shorts factors when their returns are below the historical average:

$$\operatorname{RET}_{t+1}^{\mathrm{FT}} = \frac{1}{|\mathbf{I}_t|} \sum_{i \in I_t} \operatorname{SIGN}(F_{i,t} - \overline{F}_{i,t}) F_{i,t+1},$$
(13)

where $\overline{F}_{i,t}$ is the historical average return on the *i* th factor estimated from all the data available until month *t* and sign(·) is the sign function.

As Huang et al. (2020) demonstrate, the buy-and-hold portfolio benefits from factor premiums and avoids active timing. Specifically, this portfolio longs factors with a positive prior mean return and shorts factors that earned a negative historical mean return. The portfolios are rebalanced when the sign of the historical mean changes. Huang et al. (2020) show that mean returns, rather than predictability, are the key return-generating aspects of future market momentum returns:

$$\operatorname{RET}_{i+1}^{\operatorname{BH}} = \frac{1}{|\mathbf{I}_i|} \sum_{i \in I_i} \operatorname{SIGN}(\overline{F}_{i,i}) F_{i,i+1}.$$
(14)

After constructing the factor timing and buy-and-hold strategies, we first calculate their performance. Panel A of Table 13 reports their average returns, standard deviations, Sharpe ratios, and *t*-statistics, showing that both strategies generate significant positive returns. The factor timing strategy earns an average annualized return of 10.20 % with a *t*-statistic of 2.35, while the buy-and-hold strategy produces an average return of 13.56 % (t = 3.73) per annum, significant at the 1 % level. The Sharpe ratio of the factor timing portfolio is 0.55, which is below the 0.88 ratio of the buy-and-hold portfolio. The standard deviation of the factor timing portfolio is 18.55, which is less stable than that of the buy-and-hold strategy. These findings indicate that both strategies provide considerable returns and that pure factor premiums are strong in China. The analysis with the buy-and-hold strategy reaffirms our earlier results in Panel A of Table 1 that pure factor premiums are sizable in China.

Next, we investigate whether the factor timing and buy-and-hold portfolios help explain the return of the TSFM strategy. We run the following two regressions:

$$\operatorname{RET}_{t}^{\operatorname{TSFM}} = \alpha + \beta \operatorname{RET}_{t}^{K} + \varepsilon_{t}, \tag{15}$$

$$\operatorname{RET}_{t}^{K} = \alpha + \beta \operatorname{RET}_{t}^{\operatorname{RET}} + \varepsilon_{t}, \ K \in \{\mathrm{FT}, \mathrm{BH}\},$$
(16)

where $\text{RET}_t^{\text{TSFM}}$ is the return of the TSFM strategy and RET_t^K is the return of the factor timing and buy-and-hold strategies, $K \in \{\text{FT}, \text{BH}\}$, respectively.

Panel B of Table 13 reports the results of Eq. (15). The results show that both strategies partially explain factor momentum returns. From the intercepts, factor momentum produces statistically significant and economically large alphas after controlling for the factor timing and buy-and-hold strategies, equaling 6.48 % (t = 4.76) and 6.12 % (t = 3.14), respectively. Neither strategy can subsume factor momentum returns. The coefficient on the factor timing portfolio is significant at 0.34 (t = 9.62), while that on the buy-and-hold portfolio is 0.28 with a *t*-statistic of 5.25. The loading on the factor timing portfolio is higher and more significant than that on the buy-and-hold portfolio. The R^2 of the factor timing strategy is 51.40 %, implying its ability to explain the time variation in factor momentum. The buy-and-hold portfolio has lower explanatory power ($R^2 = 24.30$ %).

Panel C of Table 13 reports the results of Eq. (16) of regressing the factor timing and buy-and-hold portfolios on the TSFM signal. The results show that both intercepts are insignificant, with -5.04 % (t = -1.49) for the factor timing portfolio and 4.80 % (t = 1.43) for the buy-and-hold portfolio. The slopes of the TSFM are positively significant for both portfolios (1.53, t = 12.54 and 0.89, t = 5.64 for the factor timing and buy-and-hold portfolios, respectively). Therefore, the TSFM signal dominates both strategies.

Overall, we find that both active timing and buy-and-hold strategies are important for the factor momentum, in contrast to the findings of Leippold and Yang (2021) in the United States. Despite the nontrivial factor premiums, conditional timing also performs

Factor Timing and Buy-and-Hold Strategies

Panel A reports the annualized average returns, standard deviations, and annualized Sharpe ratios (SR) of the factor timing and buy-and-hold strategies. The definitions of the two strategies are as follows:

$$\begin{aligned} & \textit{Ret}_{t+1}^{\textit{FT}} = \frac{1}{|\mathsf{I}_t|} \sum_{i \in I_t} \textit{sign}(F_{i,t} - \overline{F}_{i,t}) F_{i,t+1}, \\ & \textit{Ret}_{t+1}^{\textit{BH}} = \frac{1}{|\mathsf{I}_t|} \sum_{i \in I_t} \textit{sign}(\overline{F}_{i,t}) F_{i,t+1}. \end{aligned}$$

Panel B reports the results of regressing the return of the time-series factor momentum (TSFM) strategy on the factor timing and buy-and-hold portfolios separately:

 $Ret_t^{TSFM} = \alpha + \beta Ret_t^K + \varepsilon_t, \ K \in \{FT, BH\}.$

Panel C reports the results of regressing each of the returns on the factor timing and buy-and-hold portfolios on the return of the TSFM strategy separately:

 $Ret_t^K = \alpha + \beta Ret_t^{TSFM} + \varepsilon_t, \ K \in \{FT, BH\}.$

The intercepts are reported as percentages. The *t*-statistics are reported in square brackets. The sample period runs from January 2002 to December 2019.

Panel A			
	MEAN	STD	SR
FT	10.20 [2.35]	18.55	0.55
BH	13.56 [3.73]	15.41	0.88
Panel B			
	Intercept	Slope (Ret_t^K)	R ² (%)
FT	6.48 [4.76]	0.34 [9.62]	51.40
BH	6.12 [3.14]	0.28 [5.25]	24.30
Panel C			
	Intercept	Slope (Ret_t^{TSFM})	R ² (%)
FT	-5.04 [-1.49]	1.53 [12.54]	51.40
BH	4.80 [1.43]	0.89 [5.64]	24.30

well in the Chinese stock market (Tang et al., 2021). The higher volatility of factor returns, the larger proportion of retail investors in China who tend to have more behavioral biases, and the changeable market states may be the reason that timing is more important in China than in the U.S.

5.3. Influence of culture on factor momentum

Culture is an important factor that impacts investor behavior, stockholding and trading actions (Grinblatt and Keloharju, 2001), and stock price comovement (Eun et al., 2015). Kumar et al. (2011) posit that investors' religious affiliation, which is a part of a culture, can influence the construction of investment portfolios and stock returns. Chui et al. (2010) observe that cultural differences determine stock momentum returns. Specifically, markets with collectivistic cultures, like China, generate insignificant stock price momentum, while markets with individualistic cultures, like the U.S., exhibit significant price momentum. These findings inspire us to investigate the connection between culture and factor momentum.

Individuals in individualistic cultures tend to believe that their abilities are above average and are more confident in their own opinions. In contrast, people in collectivistic cultures rely more on mainstream opinions and display more herd-like behavior (Markus and Kitayama, 1991). This herding behavior might strengthen factor co-movements and, in turn, create portfolio-level momentum rather than stock-level momentum. Moreover, this behavior may increase the volatilities of factor-level returns. The Chinese stock market has a large proportion of retail investors who are more prone to herding than analyzing fundamental information. Our study reveals that factor momentum exists in a collectivistic market, the Chinese stock market. The results indicate that the relation between momentum profits and cultural differences is evident in the realm of factor momentum.

To generally examine the influence of culture on factor momentum, we construct a time-series factor momentum strategy by extracting principal component (PC) factors from the extensive set of 142 non-momentum factors proposed by Jensen et al. (2023).⁶ The formation period spans 12 months, with holding periods of 1-, 3-, 6-, and 12-months. We calculate the average annualized returns and Sharpe ratios of this TSFM-PC strategy across multiple markets, including China (CHN), Japan (JPN), Israel (IRS), Germany (DEU), Sweden (SWE), Norway (NOR), France (FRA), Denmark (DNK), Italy (ITA), the Netherlands (NLD), the United Kingdom (GBR), Australia (AUS), and the United States (USA). This PC-based TSFM encompasses a wide range of factors and helps address performance disparities of various anomalies in different markets (Kozak et al., 2020; Ehsani and Linnainmaa, 2022). The culture index is proposed by Eun et al. (2015), with smaller values signifying a collectivistic culture, and larger values denoting individualism. The results are reported in Table OA2 and Figure OA1 in Online Appendix A.

⁶ These data sets are available at https://jkpfactors.com/.

Panel B of Table OA2 shows that the TSFM-PC strategy generates significant positive returns across all markets with a 1-month holding period, affirming the robustness of factor momentum (Gupta and Kelly, 2019). The persistence of TSFM-PC is also pronounced. Only four markets display insignificant returns during longer holding periods. In Panel C of Table OA2, we find that the Sharpe ratios of TSFM-PC tend to be slightly higher in collectivistic markets compared to individualism-oriented markets. Besides, Figure OA1 illustrates a U-shaped curve for the relationship between TSFM-PC Sharpe ratios and the culture index, indicating that extreme cultural values may exert a more pronounced influence on factor momentum.

In short, the factor momentum strategy exhibits robust and significant performance "everywhere" (Gupta and Kelly, 2019). The strong and significant factor momentum is evident in both collectivistic and individualism-oriented markets, but the underlying economic mechanisms driving individualism-factor momentum and collectivistic-factor momentum may differ, and it is worth further investigation.

6. Robustness check

6.1. Different formation and holding periods

We perform our first robustness test with alternative formation and holding periods of the TSFM strategy. Specifically, we form the TSFM signal using look-back windows of one month ("1-1"), one quarter ("1-3"), two quarters ("1-6"), three quarters ("1-9"), four quarters ("1-12"), three years ("1-36"), and five years ("1-60"). We also hold the strategies for these periods. Panels A and B of Table 14 report the average annualized returns and Sharpe ratios of the long-short factor momentum portfolios based on different combinations of portfolio formation and holding periods.

Table 14 shows that the factor momentum returns are persistent. We find that 41 of the 49 alternative formation and holding period strategies generate significantly positive average returns, 35 strategies are significantly above the 5 % level, 34 strategies have an annualized return of 5 % or above, and 29 strategies produce an annualized Sharpe ratio above 0.6. The findings show the TSFM strategy's robustness and profitability.

Notably, when we use the one-month formation and holding windows, the factor momentum portfolio is also positively significant and generates an annualized return of 5.07 % with a *t*-statistic of 2.22 and a Sharpe ratio of 0.52. The results highlight that, different from the stock reversal shown in Section III. D, past factor returns still predict stronger future returns, even in the short term (Falck et al., 2022).

Additionally, when the formation period increases from one month to one year, the returns increase in China, whereas shortformation periods perform better in global markets (Gupta and Kelly, 2019). First, the Chinese stock market has a large proportion of retail investors (Jones et al., 2020), who have shorter average holding periods, exhibit greater turnover, cause short-term reversals, and attenuate momentum (Du et al., 2022). Second, the explanatory power of the significant reversal portfolio on the factor momentum strategy, reported in Section IV. A, may decrease the performance of factor momentum in China.

Overall, factor momentum is a profitable investment strategy in China, and investors may benefit from factor return autocorrelations.

6.2. Stock market crashes

The Chinese stock market was affected by the 2008 global financial crisis and the 2015 stock crash. In 2015, the Chinese stock market experienced a tremendous rise followed by a sharp drop (Liu et al., 2016). To investigate whether factor momentum exists during these two crash periods, we construct factor momentum strategies from July 2008 to December 2010 and from July 2015 to December 2017.

Table 15 shows that conditional factor momentum continues to exist and remains robust for both crash periods. Specifically, the TSFM strategy has an average return of 9.21 % (t = 2.37), and the CSFM strategy has an average return of 7.47 % (t = 2.94), similar to the main results in Table 3. The long legs contribute more to the factor momentum performance. The Sharpe ratios reach 0.93 and 1.08 for the TSFM and CSFM strategies, respectively. The factor models cannot explain the full-sample or long-leg portfolios for the conditional strategies. Overall, our findings suggest that factor momentum performs well during stock market crash periods, which is different from traditional momentum.

6.3. Price limit

China's stock market restricts daily price movements for regular stocks to a maximum of 10 %, leading to concerns about whether this regulation affects the efficacy of the factor momentum strategy. To address this issue, we remove the stocks that surpass the limit line (i.e., those that exceed 9.8 % or fall below –9.8 %) for more than 10 trading days per month. Subsequently, we construct timeseries factor momentum portfolios from January 2002 to December 2019, and present the findings in Table 16. Notably, the TSFM portfolio yields an annualized return of 9.16 % (t = 4.28) and a Sharpe ratio of 1.01, which are similar to the full sample analysis. Therefore, we conclude that stocks triggering the price limit line merely influence the overall profits of factor momentum.⁷

⁷ We also test the effect of price limit on stock momentum and find that after deleting the stocks that trigger the price limit line, the performance of stock momentum in China increases. Its Sharpe ratio reaches 0.41.

Different Holding and Formation Periods

Panels A and B report the annualized average returns and *t*-statistics (in square brackets) as well as the annualized Sharpe ratios of the alternative holding and formation periods of the time-series factor momentum strategy of the 10 non-momentum factors in China, respectively. Specifically, the factor momentum signals are constructed based on the average return in the prior *k* months and held for *h* months with *k*, h = 1, 3, 6, 9, 12, 36, and 60, without dropping the most recent month. We bold all the positively significant returns. All returns are reported in percentage. The sample period runs from January 2002 to December 2019.

Holding/Formation	1	3	6	9	12	36	60
Panel A: Mean Return							
1	5.07 [2.22]	5.74 [2.47]	4.26 [1.79]	6.85 [3.42]	9.91 [4.88]	8.61 [4.14]	6.92 [3.60]
3	3.18 [1.35]	4.91 [2.12]	3.91 [1.72]	7.47 [3.54]	7.64 [3.55]	8.98 [4.28]	6.34 [3.34]
6	-0.12 [-0.05]	3.16 [1.36]	5.69 [2.48]	6.02 [2.78]	6.83 [3.26]	6.74 [3.21]	5.85 [2.96]
9	1.11 [0.51]	5.54 [2.71]	5.90 [2.89]	9.26 [4.69]	8.22 [4.03]	9.19 [4.82]	5.81 [3.09]
12	3.50 [1.49]	4.77 [2.24]	5.84 [2.66]	6.59 [3.27]	7.49 [3.55]	7.35 [3.29]	6.24 [2.91]
36	4.01 [1.75]	6.80 [3.04]	5.84 [2.48]	7.70 [3.45]	8.51 [3.63]	8.29 [3.77]	6.03 [2.70]
60	-6.87 [-2.56]	-0.76 [-0.30]	2.30 [0.80]	4.78 [1.67]	5.15 [1.77]	4.61 [1.84]	5.20 [2.17]
Panel B: Sharpe Ratio							
1	0.52	0.58	0.42	0.80	1.15	0.97	0.85
3	0.32	0.50	0.41	0.83	0.83	1.00	0.79
6	-0.01	0.32	0.58	0.65	0.77	0.75	0.70
9	0.12	0.64	0.68	1.10	0.95	1.13	0.73
12	0.35	0.53	0.63	0.77	0.83	0.77	0.68
36	0.43	0.75	0.61	0.85	0.89	0.92	0.66
60	-0.67	-0.08	0.21	0.44	0.46	0.48	0.57

Table 15

Performance of the TSFM and CSFM Strategies in Stock Market Crash Periods

This table reports the annualized average returns, standard deviations, Sharpe ratios (SR), abnormal alphas (FF5- α , CH3- α , and Cond-CH3- α), and corresponding *t*-statistics (in square brackets) for time-series and cross-sectional factor momentum, winners, and losers with the 10 non-momentum factors in China's two market crash periods. The conditional CH3 model (Cond-CH3) imposes the investor sentiment into the model's β . All returns are reported in percentage. The sample periods run from July 2008 to December 2010 and from July 2015 to December 2017.

	Mean	STD	SR	FF5-α	СНЗ-а	Cond-CH3-α
Panel A: Time-series fa	ctor momentum					
Full Sample	9.21 [2.37]	9.90	0.93	9.84 [2.43]	8.64 [1.86]	8.51 [1.84]
Winners	19.49 [5.06]	11.53	1.69	14.40 [5.71]	8.76 [2.38]	8.62 [2.33]
Losers	12.54 [1.97]	16.50	0.76	8.16 [1.52]	3.00 [0.56]	3.02 [0.55]
Panel B: Cross-sectional factor momentum						
Full Sample	7.47 [2.94]	6.92	1.08	6.48 [2.36]	5.04 [1.88]	4.96 [1.82]
Winners	24.26 [5.05]	14.79	1.64	17.76 [4.73]	11.76 [2.68]	11.60 [2.63]
Losers	9.31 [2.25]	10.95	0.85	4.68 [1.63]	1.56 [0.40]	1.42 [0.38]

Table 16

Performance of the TSFM Considering Price Limit

This table reports the annualized average returns, standard deviations, Sharpe ratios (SR), and corresponding *t*-statistics (in square brackets) for time-series factor momentum, winners, and losers with the 10 non-momentum factors considering the price limitation. Specifically, we delete the stocks that trigger the limit line (above 9.8 % or below –9.8 %) more than 10 trading days in a month. All returns are reported in percentage. The sample period runs from January 2002 to December 2019.

	Mean	STD	SR
Full Sample	9.16 [4.28]	9.09	1.01
Winners	9.06 [4.10]	9.39	0.96
Losers	-8.94 [-3.37]	11.25	-0.79

7. Conclusion

Factor momentum strategies are valuable, beneficial, and robust in the Chinese stock market. By employing 10 commonly used non-momentum factors, we demonstrate that both the TSFM and CSFM strategies generate positively significant returns and high Sharpe ratios. The TSFM strategy outperforms the CSFM strategy, as the former purely bets on the factor return autocorrelations.

We document the strong power of factor momentum in subsuming stock momentum, high-priced stock momentum, and industry momentum, along with digesting its constituent factors as well as various categories of anomalies. Stock-level momentum and reversal strategies cannot absorb factor momentum. Furthermore, we examine the economic channels of factor momentum. We find that factor

T. Ma et al.

momentum generates higher returns during periods of higher aggregate idiosyncratic volatility and lower investor sentiment as well as among stocks with higher information asymmetry and short-sale constraints, consistent with increased arbitrageur participation in correcting mispricing under these scenarios. Moreover, we observe that factor timing and factor premiums both contribute to factor momentum returns. In addition, the collectivistic culture, that leading to herding behavior, may contribute to the factor momentum.

This study thus answers the question of whether factor momentum is statistically and economically significant in a market that lacks stock-level momentum, such as the Chinese stock market. The explanatory power and economic grounds of factor momentum in China provide important implications for the global market.

Author statement

All authors have equal contribution.

Supplementary materials

Supplementary material associated with this article can be found, in the online version, at doi:10.1016/j.jempfin.2023.101458.

References

Abreu, D., Brunnermeier, M.K., 2002. Synchronization risk and delayed arbitrage. J. Financ. Econ. 66 (2-3), 341-360. Allen, F., Oian, J., Shan, C., Zhu, J., 2020. The development of the Chinese stock market. In: Amstad, M., Sun, G., Xiong, W. (Eds.), Handbook of China's Financial System. Princeton University Press, pp. 283-313. Amihud, Y., 2002. Illiquidity and stock returns: cross-section and time-series effects. J. Financ. Mark. 5 (1), 31-56. Arnott, R.D., Clements, M., Kalesnik, V., & Linnainmaa, J.T. (2021). Factor momentum. Available at SSRN 3116974. Asness, C.S., Porter, R.B., & Stevens, R.L. (2000). Predicting stock returns using industry-relative firm characteristics. Available at SSRN 213872. Avramov, D., Cheng, S., Schreiber, A., Shemer, K., 2017. Scaling up market anomalies. J. Investing 26 (3), 89–105. Baker, M., Wurgler, J., 2006. Investor sentiment and the cross-section of stock returns. J. Finance 61 (4), 1645–1680. Banz, R.W., 1981. The relationship between return and market value of common stocks. J. Financ. Econ. 9 (1), 3-18. Basu, S., 1983. The relationship between earning's yield, market value and return for NYSE common stocks: further evidence. J. Financ. Econ. 12 (1), 129–156. Boubaker, S., Du, L., Liu, Z., 2021. Industry momentum with correlation consolidation: evidence from China. J. Asset Manag. 1-10. Boubaker, S., Mansali, H., Rjiba, H., 2014. Large controlling shareholders and stock price synchronicity. J. Bank Financ. 40, 80–96. Burnside, C., Eichenbaum, M., Rebelo, S., 2011. Carry trade and momentum in currency markets. Ann. Rev. Financ. Econ. 3 (1), 511-535. Carpenter, J.N., Lu, F., Whitelaw, R.F., 2021. The real value of China's stock market. J. Financ. Econ. 139 (3), 679-696. Cheema, M.A., Nartea, G.V., 2017. Momentum returns, market states, and market dynamics: is China different? Int. Rev. Econ. Finance 50, 85-97. Chen, Y., Zhao, H., 2012. Informed trading, information uncertainty, and price momentum. J. Bank Financ. 36 (7), 2095–2109. Chui, A.C., Titman, S., Wei, K.J., 2010. Individualism and momentum around the world. J. Finance 65 (1), 361-392. Cohen, R.B., & Polk, C.K. (1998). The impact of industry factors in asset-pricing tests, Kellogg Graduate School of Management working paper. Daniel, K., Moskowitz, T.J., 2016. Momentum crashes. J. Financ. Econ. 122 (2), 221-247. Daniel, K., Hirshleifer, D., Subrahmanyam, A., 1998. Investor psychology and security market under-and overreactions. the. J. Finance 53 (6), 1839–1885. Datar, V.T., Naik, N.Y., Radcliffe, R., 1998. Liquidity and stock returns: an alternative test. J. Financ. Mark. 1 (2), 203-219. Dierkes, M., Krupski, J., 2022. Isolating momentum crashes. J. Empiric. Finance 66, 1-22. Dong, X., Kang, N., & Peress, J. (2020). Fast and slow arbitrage: fund flows and mispricing in the frequency domain. CEPR Discussion Paper No. DP15235. Dong, X., Li, Y., Rapach, D.E., Zhou, G., 2022. Anomalies and the expected market return. J. Finance 77 (1), 639-68. Du, J., Huang, D., Liu, Y.-J., Shi, Y., Subrahmanyam, A., & Zhang, H. (2022), Retail investors and momentum. Available at SSRN 4163257. Duffie, D., 2010. Presidential address: asset price dynamics with slow-moving capital. J. Finance 65 (4), 1237–1267. Ehsani, S., Linnainmaa, J.T., 2022. Factor momentum and the momentum factor. J. Finance 77, 1877–1919. Engelberg, J., McLean, R.D., Pontiff, J., Ringgenberg, M.C., 2021. Do cross-sectional predictors contain systematic information? J. Financ. Quant. Anal. Forthcoming. Eun, C.S., Wang, L., Xiao, S.C., 2015. Culture and R2. J. Financ. Econ. 115 (2), 283-303. Falck, A., Rej, A., Thesmar, D., 2022. Is factor momentum greater than stock momentum? J. Invest. Strat. 10 (4). Fama, E.F., French, K.R., 2015. A five-factor asset pricing model. J. Financ. Econ. 116 (1), 1-22. Frazzini, A., Pedersen, L.H., 2014. Betting against beta. J. Financ. Econ. 111 (1), 1-25. Gao, Y., Guo, B., Xiong, X., 2021a. Signed momentum in the Chinese stock market. Pac.-Basin Finance J. 68. Gao, Y., Han, X., Li, Y., Xiong, X., 2021b. Investor heterogeneity and momentum-based trading strategies in China. Int. Rev. Financ. Anal. 74, 101654. Garcia, R., Mantilla-García, D., Martellini, L., 2014. A model-free measure of aggregate idiosyncratic volatility and the prediction of market returns. J. Financ. Quant. Anal. 49 (5-6), 1133-1165. Gårleanu, N., Pedersen, L.H., 2013. Dynamic trading with predictable returns and transaction costs. J. Finance 68 (6), 2309-2340. Gârleanu, N., Pedersen, L.H., 2016. Dynamic portfolio choice with frictions. J. Econ. Theory 165, 487-516. George, T.J., Hwang, C.Y., 2004. The 52-week high and momentum investing. J. Finance 59 (5), 2145-2176. Green, J., Hand, J.R., Zhang, X.F., 2017. The characteristics that provide independent information about average US monthly stock returns. Rev. Financ. Stud. 30 (12), 4389-4436 Grinblatt, M., Han, B., 2005. Prospect theory, mental accounting, and momentum. J. Financ. Econ. 78 (2), 311-339. Grinblatt, M., Kelohariu, M., 2001, How distance, language, and culture influence stockholdings and trades. J. Finance 56 (3), 1053–1073. Gu, S., Kelly, B., Xiu, D., 2020. Empirical asset pricing via machine learning. Rev. Financ. Stud. 33 (5), 2223-2273. Gupta, T., Kelly, B., 2019. Factor momentum everywhere. J. Portfolio Manag. 45 (3), 13-36. Gutierrez Jr, R.C., Kelley, E.K, 2008. The long-lasting momentum in weekly returns. J. Finance 63 (1), 415-447. Haddad, V., Kozak, S., Santosh, S., 2020. Factor timing. Rev. Financ. Stud. 33 (5), 1980-2018. Heston, S., Jones, C.S., Khorram, M., Li, S., Mo, H., 2021. Option Momentum. J. Finance. Forthcoming. Hong, H., Stein, J.C., 2003. Differences of opinion, short-sales constraints, and market crashes. Rev. Financ. Stud. 16 (2), 487-525. Hou, K., Xue, C., Zhang, L., 2020. Replicating anomalies. Rev. Financ. Stud. 33 (5), 2019–2133. Huang, D., Li, J., Wang, L., Zhou, G., 2020. Time series momentum: is it there? J. Financ. Econ. 135 (3), 774-794. Jegadeesh, N., Titman, S., 1993. Returns to buying winners and selling losers: implications for stock market efficiency. J. Finance 48 (1), 65-91. Jegadeesh, N., Titman, S., 2001. Profitability of momentum strategies: an evaluation of alternative explanations. J. Finance 56 (2), 699–720.

Jensen, T.I., Kelly, B., Pedersen, L.H., 2023. Is there a replication crisis in finance? J. Finance 1-54.

Jin, L., Myers, S.C., 2006. R² around the World: new theory and new tests. J. Financ. Econ. 79 (2), 257-292. Jones, C.M., Shi, D., Zhang, X., & Zhang, X. (2020). Heterogeneity in retail investors: evidence from comprehensive account-level trading and holdings data. Available at SSRN, 3628809.

Kozak, S., Nagel, S., Santosh, S., 2020. Shrinking the cross-section. J. Financ. Econ. 135 (2), 271-292.

Kumar, A., Page, J.K., Spalt, O.G., 2011. Religious beliefs, gambling attitudes, and financial market outcomes. J. Financ. Econ. 102 (3), 671-708.

Leippold, M., & Yang, H. (2021). The anatomy of factor momentum. Available at SSRN, 3517888.

Leippold, M., Wang, Q., Zhou, W., 2021. Machine learning in the Chinese stock market. J. Financ. Econ.

Lewellen, J., 2002. Momentum and autocorrelation in stock returns. Rev. Financ. Stud. 15 (2), 533-564.

Li, S.Z., Yuan, P., & Zhou, G. (2022). Risk-based momentum. Available at SSRN, 4062260.

Lin, Q., 2022. Understanding idiosyncratic momentum in the Chinese stock market. J. Int. Financ. Mark., Inst. Money 76, 101469.

Liu, D., Gu, H., Xing, T., 2016. The meltdown of the Chinese equity market in the summer of 2015. Int. Rev. Econ. Finance 45, 504-517.

Liu, J., Stambaugh, R.F., Yuan, Y., 2019. Size and value in China. J. Financ. Econ. 134 (1), 48-69. Lo, A.W., MacKinlay, A.C., 1990. When are contrarian profits due to stock market overreaction? Rev. Financ. Stud. 3 (2), 175-205.

Markus, H.R., Kitayama, S., 1991. Culture and the self: implications for cognition, emotion, and motivation. Psychol. Rev. 98 (2), 224.

Menkhoff, L., Sarno, L., Schmeling, M., Schrimpf, A., 2012. Currency momentum strategies. J. Financ. Econ. 106 (3), 660-684.

Miller, E.M., 1977. Risk, uncertainty, and divergence of opinion. J. Finance 32 (4), 1151-1168.

Morck, R., Yeung, B., Yu, W., 2000. The information content of stock markets: why do emerging markets have synchronous stock price movements? J. Financ. Econ. 58 (1-2), 215-260.

Moskowitz, T.J., Grinblatt, M., 1999. Do industries explain momentum? J. Finance 54 (4), 1249–1290.

Moskowitz, T.J., Ooi, Y.H., Pedersen, L.H., 2012. Time series momentum. J. Financ. Econ. 104 (2), 228-250.

Novy-Marx, R., 2012. Is momentum really momentum? J. Financ. Econ. 103, 429-453.

Pan, L., Tang, Y., Xu, J., 2013. Weekly momentum by return interval ranking. Pac.-Basin Finance J. 21 (1), 1191–1208.

Rachev, S., Jašić, T., Stoyanov, S., Fabozzi, F.J., 2007. Momentum strategies based on reward-risk stock selection criteria. J. Bank. Finance 31, 2325-2346.

Roll, R., 1988. R-squared. J. Finance 43, 541-566.

Rosenberg, B., Reid, K., Lanstein, R., 1985. Persuasive evidence of market inefficiency. J. Portfolio Manag. 11, 9-16.

Sihvonen, M. (2021). Yield curve momentum. Available at SSRN 3965229.

Sloan, R.G., 1996. Do stock prices fully reflect information in accruals and cash flows about future earnings? Account, Rev. 289-315.

Stambaugh, R.F., Yu, J., Yuan, Y., 2012. The short of it: investor sentiment and anomalies. J. Financ. Econ. 104 (2), 288-302.

Tang, G., Jiang, F., Qi, X., Huang, N., 2021. It takes two to tango: fundamental timing in stock market. Int. J. Finance Econ. 26 (4), 5259-5277.

Titman, S., Wei, K.J., Xie, F., 2004. Capital investments and stock returns. J. Financ. Quant. Anal. 39 (4), 677-700.

Zaremba, A., Shemer, J., 2018. Is there momentum in factor premia? Evidence from international equity markets, Res. Int. Bus. Finance 46, 120–130.

Zhang, S., 2022. Dissecting currency momentum. J. Financ. Econ. 144 (1), 154–173.

Zhang, X.F., 2006. Information uncertainty and stock returns. J. Finance 61 (1), 105-137.